

© 2021 ESO

Document downloaded from the institutional repository of the University of Alcala: <u>http://ebuah.uah.es/dspace/</u>

This is a posprint version of the following published document:

Rodríguez García, L., Gómez Herrero, R. [et al.], 2021, "The unusual widespread solar energetic particle event on 2013 August 19: Solar origin and particle longitudinal distribution", Astronomy & Astrophysics, vol. 653, art. no. A137, pp. 1-22.

Available at https://dx.doi.org/10.1051/0004-6361/202039960

(Article begins on next page)



This work is licensed under a

Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License.

The unusual widespread solar energetic particle event on 2013 August 19

Solar origin and particle longitudinal distribution

L. Rodríguez-García¹, R. Gómez-Herrero¹, I. Zouganelis², L. Balmaceda^{3,4}, T. Nieves-Chinchilla³, N. Dresing^{5,6}, M. Dumbović⁷, N. V. Nitta⁸, F. Carcaboso¹, L. F. G. dos Santos⁹, L. K. Jian³,

L. Mays³, D. Williams², and J. Rodríguez-Pacheco¹

¹ Universidad de Alcalá, Space Research Group, Alcalá de Henares, Madrid, Spain

e-mail: l.rodriguezgarcia@edu.uah.es

² European Space Astronomy Center, European Space Agency, Villanueva de la Cañada, Madrid, Spain

³ Heliophysics Science Division, NASA Goddard Space Flight Center, Greenbelt, MD, USA

⁴ George Mason University, Fairfax, VA, USA

⁵ Institut fuer Experimentelle und Angewandte Physik, University of Kiel, Kiel, Germany

⁶ Department of Physics and Astronomy, University of Turku, FI-20014 Turku, Finland

⁷ Hvar Observatory, Faculty of Geodesy, University of Zagreb, Croatia

⁸ Lockheed Martin Solar and Astrophysics Laboratory, Palo Alto, CA, USA

⁹ CIRES, University of Colorado Boulder, Boulder, CO, USA

Received November 22, 2020; accepted June 23, 2021

ABSTRACT

Context. Late on 2013 August 19, STEREO-A, STEREO-B, MESSENGER, Mars Odyssey, and the L1 spacecraft, spanning a longitudinal range of 222° in the ecliptic plane, observed an energetic particle flux increase. The widespread solar energetic particle (SEP) event was associated with a coronal mass ejection (CME) that came from a region located near the far-side central meridian from Earth's perspective. The CME erupted in two stages, and was accompanied by a late M-class flare observed as a post-eruptive arcade, persisting low-frequency (interplanetary) type II and groups of shock-accelerated type III radio bursts, all of them making this SEP event unusual.

Aims. There are two main objectives of this study, disentangling the reasons for the different intensity-time profiles observed by the spacecraft, especially at MESSENGER and STEREO-A locations, longitudinally separated by only 15°, and unravelling the single solar source related with the widespread SEP event.

Methods. The analysis of in situ data, such as particle fluxes, anisotropies and timing, and plasma and magnetic field data, is compared with the remote-sensing observations. A spheroid model is applied for the CME-driven shock reconstruction and the ENLIL model is used to characterize the heliospheric conditions, including the evolution of the magnetic connectivity to the shock.

Results. The solar source associated with the widespread SEP event is the shock driven by the CME, as the flare observed as a posteruptive arcade is too late to explain the estimated particle onset. The different intensity-time profiles observed by STEREO-A, located at 0.97 au, and MESSENGER, at 0.33 au, can be interpreted as enhanced particle scattering beyond Mercury's orbit. The longitudinal extent of the shock does not explain by itself the wide spread of particles in the heliosphere. The particle increase observed at L1 may be attributed to cross-field diffusion transport, and this is also the case for STEREO-B, at least until the spacecraft is eventually magnetically connected to the shock when it reaches ~0.6 au.

Key words. Sun: particle emission- Sun: coronal mass ejections (CMEs) - Sun: corona - Sun: heliosphere

1. Introduction

Solar activity provides the origin and environment for solar energetic particle (SEP) events, which are significant increases in the fluxes of charged particles measured in situ, mainly protons and electrons, being ejected from the Sun (Reames 2013). These SEP events are associated with coronal mass ejections (CMEs) and X-ray flares, and all three are the result of the release of the huge energy stored in the magnetic field structures formed at the Sun. The release, acceleration, and transport of particles, and the relation between CMEs, flares, and SEP events have been studied over several decades (Forbush 1946; Cane et al. 1986, 1988) and are still under investigation (e.g. Lario et al. 2013, 2014, 2017; Kihara et al. 2020, and references therein). The SEP events are often classified into two categories, impulsive and gradual (Cane et al. 1986; Reames 1999), on account of their observed properties, such as timescales, spectra, composition and charge states, and the associated radio bursts. However, some observations (Kocharov & Torsti 2002; Kallenrode 2003; Papaioannou et al. 2016) indicate that there might be no distinct separation between the two groups, and they suggest that a continuum of event properties is present, including hybrid or mixed SEP events. According to the two-class paradigm, impulsive SEP events are primarily accelerated by processes related with flares and jets, and gradual SEP events are the result of particle acceleration in the corona and the solar wind by CMEdriven shock waves. If the CMEs and their interplanetary (IP) counterparts, hence IP CMEs (ICMEs), are fast enough, the corresponding coronal and IP shocks that they drive can accelerate SEPs to high energies (e.g. Desai & Giacalone 2016, and references therein). Thus, gradual SEP events tend to be intense, energetic, and spatially and temporally extended, reflecting particles that have been accelerated over long timescales (Reames 2013).

During some gradual events, SEPs originating from the CME-driven shock are detected over a very wide range of heliolongitudes. These widespread SEP events have been extensively researched (e.g. Reames et al. 1996; Wibberenz & Cane 2006; Lario et al. 2006, 2013, 2016; Dresing et al. 2012, 2014; Papaioannou et al. 2014; Richardson et al. 2014; Gómez-Herrero et al. 2015; Paassilta et al. 2018; Xie et al. 2019) thanks to constellations of spacecraft widely distributed throughout the heliosphere, such as Helios 1, Helios 2, Ulysses, the SOlar and Heliographic Observatory (SOHO, Domingo et al. 1995), the Solar TErrestrial RElations Observatory (STEREO, Kaiser et al. 2008), and other planetary missions. The ever-changing configuration of the STEREO spacecraft is advantageous for the study of the longitudinal variations of SEP events near 1 au, together with, for instance, SOHO located at L1. Several studies have clearly demonstrated that SEPs can arrive at Earth from solar events anywhere on the far side of the Sun, despite their great distance from the footpoint of the nominal spiral magnetic field line connected to Earth (e.g. Dodson & Hedeman 1969; Torsti et al. 1999; Cliver et al. 2005; Dresing et al. 2012; Richardson et al. 2014; Gómez-Herrero et al. 2015).

The particular evolving connection of the observer to the CME-driven shock results in different SEP intensity profiles (e.g. Cane et al. 1988; Reames 1999; Cane & Lario 2006). It is also known that a driven shock is often fastest and strongest near the 'nose', while it is expected to show a decrease in shock speed and strength towards the 'flank' areas (Kallenrode et al. 1993; Neugebauer 2013). The term nose of the shock is defined in opposition to the flanks, but it may lead to an over-simplistic depiction of the complexity, width, and size of the shock surface (Reames 2013). Therefore, the knowledge of the CME and CME-driven shock parameters, such as geometry and speed, plays a key role in understanding SEP events (e.g. Kallenrode & Wibberenz 1997; Lario et al. 2013; Desai & Giacalone 2016; Kouloumvakos et al. 2019). The angular extent of IP shocks can be investigated using multi-point in situ data, while the size of the coronal shocks can be indirectly deduced via remote-sensing observations. However, as the information provided by a single coronagraph image is an approximation of the real structure, the multi-view 3D reconstruction of the coronal shock geometry to correct projection effects is preferable (e.g. Lario et al. 2014, 2016).

The extent of the coronal and IP shocks may sometimes explain the extremely wide longitudinal spread of particles (e.g. Lario et al. 2016). However, in other SEP events, some additional physical processes allowing particle propagation perpendicularly to the average direction of the IP magnetic field (IMF) are required to explain the observations. These processes could include scattering by IP magnetic field turbulence (e.g. Dresing et al. 2012; Dröge et al. 2016), field line meandering (e.g. Laitinen et al. 2015, and references therein), and deviations from the ideal IP magnetic field line spiral structure (e.g. Richardson et al. 1991; Richardson & Cane 1996; Gómez-Herrero et al. 2007; Chollet et al. 2010).

As an example of deviations from ideal IP magnetic field configuration, CMEs prior to an SEP event can distort the nominal transport conditions between the Sun and the different space-

Article number, page 2 of 23

craft (e.g. Lario & Karelitz 2014). To have an insight into the possible solar wind structures present in the heliosphere in which CMEs and SEPs propagate, preceding eruptions generally have to be included in global 3D magnetohydrodynamic (MHD) simulations (e.g. Bain et al. 2016; Dumbović et al. 2019). This characterization of the heliosphere is more accurate when the background solar wind is well replicated and if multi-point coronagraph observations are used to infer CME parameters (e.g. Lee et al. 2013; Mays et al. 2015).

On 2013 August 19, an unusual large and widespread SEP event occurred that could be seen by different spacecraft located in the inner heliosphere, spanning a longitudinal range of 222° in the ecliptic plane. The SEP event was observed by *MErcury Surface Space ENvironment GEochemistry and Ranging* (MES-SENGER, Solomon et al. 2007), STEREO-A, and *Mars Odyssey* (Saunders et al. 2004), all three being nominally well connected to the source, and also by STEREO-B and spacecraft at L1, such as SOHO and the *Advanced Composition Explorer* (ACE, Stone et al. 1998), all having large longitudinal separation between the solar source and the footpoint of the respective nominal field lines. The SEP origin was associated with a wide and fast CME, erupting near the far-side central meridian from Earth's perspective.

The term 'unusual' is linked to several circumstances. Firstly, the single CME, driving a single shock associated with the SEP event, erupted in two stages. Secondly, the observed radio dynamic spectrum associated with the SEP event presented groups of shock-accelerated (SA) type III bursts from \sim 50 minutes before to \sim 40 minutes after the release of particles. Thirdly, the flare observed as a post-eruptive arcade was not likely related with the particle acceleration.

In addition to the unusual features, the difference in intensity profiles observed by STEREO-A (delayed and gradual) and MESSENGER (prompt and sharp), which were both relatively well connected to the source and only separated by 15° in heliolongitude, provides a good opportunity for the analysis of the SEP IP propagation between ~0.3 au and ~1 au.

To shed some light on which physical mechanisms are behind this unusual widespread SEP event, there are two specific objectives in this study: firstly, to identify the origin and evolution of the single solar source related with the widespread SEP event; and secondly, to disentangle the acceleration and propagation conditions that might be responsible for the SEP time profiles observed at different spacecraft, especially at the locations of MESSENGER and STEREO-A.

Section 2 introduces the instrumentation used in this paper. Section 3 presents the SEP event, showing the energetic particle fluxes, anisotropies, spectra and timing at different locations, as well as the magnetic connectivity. Section 4 examines the SEP parent solar source, including remote-sensing observations, such as extreme ultraviolet (EUV), coronagraph, soft-X ray, and radio emission. It also presents the reconstruction of the CME, using the graduated cylindrical shell (GCS) model developed by Thernisien et al. (2006) and Thernisien (2011), and the associated CME-driven shock, applying an spheroid model based on Olmedo et al. (2013). Section 5 presents the heliospheric conditions that might affect the particle transport. We analyse the in situ solar wind along with the Wang-Sheeley-Arge (WSA)-ENLIL + Cone model (hereafter ENLIL model; Odstrčil et al. 1996; Arge & Pizzo 2000; Odstrcil 2003; Arge et al. 2004) simulation. Section 6 summarizes the SEP event and presents the findings of the research. Finally, Sect. 7 outlines the main conclusions.



Fig. 1. Overview of the widespread SEP event observed on 2013 August 19 at different spacecraft locations. (a) View from the north heliographic pole showing the locations of STEREO-A (red), near-Earth observers (L1; green), STEREO-B (blue), MESSENGER (brown) and Mars Odyssey (purple) on 2013 August 20. R and Long indicate the heliocentric distance and heliographic longitude of each observer, respectively, in the HEEQ coordinates. Also shown are the nominal IP magnetic field lines connecting each spacecraft with the Sun (yellow circle at the centre, not to scale), considering the mean solar wind measured around the solar release time of the SEP event (for MESSENGER and Mars Odyssey the same speed as measured at STEREO-A was assumed). The orange line indicates the longitude of the parent CME apex direction. (b) Particle counting rate measured by Mars Odyssey in the higher channels of the inner scintillator with the background subtracted and smoothed with a third-order polynomial regression. (c) Near-relativistic electron intensities measured by MESSENGER. (d), (e), and (f) Near-relativistic electron (top) and proton intensities measured by STEREO-A, STEREO-B, and ACE and SOHO, respectively. The orange arrows on the horizontal axis indicate the occurrence time of the parent CME, as observed in EUV data, and the vertical orange lines show the passage of the IP shocks related with the SEP event. The shocks arriving the Earth that are not associated with the parent solar source are indicated with vertical grey dashed lines in (f). The horizontal lines in (d) and (e) represent a period with significant proton contamination of the electron fluxes observed by both STEREO/SEPT.

2. Instrumentation

The comprehensive study of the SEP event requires the analysis of observations from a wide range of instrumentation on board different spacecraft. We used data from STEREO-A, STEREO-B, MESSENGER, ACE, SOHO, Mars Odyssey, Wind (Szabo 2015), the *Solar Dynamics Observatory* (SDO, Pesnell et al. 2012), the *Geostationary Operational Environmental Satellites* (GOES, Garcia 1994), and also from ground-based experiments, such as the *Learmonth spectrograph*¹ and the *Bruny Island Ra*-

dio Spectrometer (BIRS, Erickson 1997). Remote-sensing observations of the CME and the activity on the solar surface were provided by the Atmospheric Imaging Assembly (AIA, Lemen et al. 2012) on board SDO, the C2 and C3 coronagraphs of the Large Angle and Spectrometric COronagraph (LASCO, Brueckner et al. 1995) instrument on board SOHO, and the Sun Earth Connection Coronal and Heliospheric Investigation (SEC-CHI, Howard et al. 2008) instrument suite on board STEREO. In particular, the COR1 and COR2 coronagraphs and the Extreme Ultraviolet Imager (EUVI, Wuelser et al. 2004), part of SECCHI suite, were utilized. Radio observations were provided

http://www.sws.bom.gov.au/Solar/3/2

A&A proofs: manuscript no. output

Spacecraft location				Solar Wind	Field line	Field line footpoint location			Connection	
		Speed	Length					Angle		
]	Location	a	In situ/ENLIL ^b	Parker/ENLIL ^d	Parker Spiral ^e ENLIL ^f		LIL^{f}	Parker/ENLIL ^g	
s/c	R	Lon	Lat	Vsw	L	Lon	Lat	Lon	Lat	$\Delta \phi$
	(au)	(deg)	(deg)	(km/s)	(au)	(deg)	(deg)	(deg)	(deg)	(deg)
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
STEREO-B	1.02	-138	-7	332/377	1.24/1.26	-64	-7	-87	-6	125/102
MESSENGER	0.33	159	1	418°/301	0.31/0.31	177	1	174	1	6/3
STEREO-A	0.97	144	-4	418/315	1.08/1.14	-160	-4	-164	-5	29/25
MARS	1.58	128	-1	418°/322	2.09/2.26	-141	-1	-145	-3	48/44
EARTH	1.01	0	7	446/477	1.12/1.09	54	7	33	6	-117/-138

Notes.

^a STEREO location in HEEQ coordinates at 23:22 universal time (UT) on 2013 August 19. MARS and MESSENGER location in HEEQ coordinates at 23:00 UT on 2013 August 19.

^b One-hour averaged solar wind speed measured at estimated particle solar release time. ENLIL values are taken from simulation on 2013 August 19 at 23:00 UT.

^c Mercury and Mars solar wind speeds from STEREO-A.

^d Length calculated from 6 (5) R_{\odot} for Parker spiral (ENLIL) magnetic field lines (as the probable heliospheric particle release height), assuming a radial extension from 21.5 R_{\odot} to 5 R_{\odot} in the case of ENLIL modelling.

^e Footpoint location at 6 R_{\odot} .

^f Footpoint location radially extended from 21.5 R_{\odot} to 5 R_{\odot} .

^g The positive sign means westward location with respect to the CME apex direction (171°).

by the S/WAVES (Bougeret et al. 2008) investigation on board STEREO, the WAVES (Bougeret et al. 1995) experiment on board Wind, and the Learmonth and the BIRS spectrometers. The X-Ray telescopes of the GOES satellites were also used. In situ energetic particle observations were provided by the Solar Electron and Proton Telescope (SEPT, Müller-Mellin et al. 2008), the Low-Energy Telescope (LET, Mewaldt et al. 2008), and the High-Energy Telescope (HET, von Rosenvinge et al. 2008) on board STEREO (all of them part of the IMPACT instrument suite, Luhmann et al. 2008); the Electron Proton and Alpha Monitor (EPAM, Gold et al. 1998) on board ACE; the Electron Proton Helium INstrument (EPHIN), part of the Comprehensive Suprathermal and Energetic Particle Analyzer (COSTEP, Müller-Mellin et al. 1995) and the Energetic Relativistic Nuclei and Electron Instrument (ERNE, Torsti et al. 1995) on board SOHO; the 3D Plasma and Energetic Particle Investigation (3DP)² on board Wind; the Energetic Particle and Plasma Spectrometer (EPPS, Andrews et al. 2007) on board MESSENGER; and the High Energy Neutron Detector (HEND), part of the Gamma Ray Spectrometer Suite (GRS, Boynton et al. 2004), on board Mars Odyssey, which is sensitive to charged particles although it was not primarily designed to measure them (e.g. Zeitlin et al. 2010). Solar wind plasma and magnetic field observations were provided by the Plasma and Suprathermal Ion Composition (PLASTIC, Galvin et al. 2008) investigation and the Magnetic Field Experiment (Acuña et al. 2008) on board STEREO; the Magnetometer Instrument (Anderson et al. 2007) on board MESSENGER; and the Magnetic Field Experiment (Smith et al. 1998), the Solar Wind Electron Proton Alpha Monitor (SWEPAM, McComas et al. 1998) and

the Solar Wind Ion Composition Spectrometer (SWICS, Gloeckler et al. 1998) on board ACE. The CDAW SOHO LASCO CME catalogue³ (Yashiro et al. 2004) provided the CME timing from Earth's perspective. Interplanetary structures were classified using the STEREO level 3 event lists⁴ (Jian et al. 2018, 2019) maintained by L. Jian, the near-Earth ICME list provided by I. Richardson and H. Cane⁵ (Richardson & Cane 2010), the SOHO CELIAS/MTOF Proton Monitor shock list⁶, the IP shocks catalogue maintained by the University of Helsinki⁷, and the ICME catalogue at Mercury from the University of New Hampshire⁸. The magnetograms from Global Oscillations Network Group (GONG, Harvey et al. 1996) are available from the National Solar Observatory website⁹.

3. SEP event: In situ observations and analysis

By the end of 2013 August 19 and beginning of August 20, STEREO-A, STEREO-B, and L1 saw a moderate and gradual increase in both near-relativistic electrons, from 65 to 105 keV, and in protons up to 50 MeV energies, while MESSENGER detected a sharp and prompt increase in near-relativistic electrons up to the 708 keV channel, and so did Mars-Odyssey, at least in protons energies above 27 MeV; the details about the

² http://sprg.ssl.berkeley.edu/wind3dp/

³ https://cdaw.gsfc.nasa.gov/CME_list/

⁴ https://stereo-ssc.nascom.nasa.gov/data/ins_data/ impact/level3/

⁵ http://www.srl.caltech.edu/ACE/ASC/DATA/level3/ icmetable2.htm

⁶ http://umtof.umd.edu/pm/FIGS.HTML

⁷ http://www.ipshocks.fi/

⁸ http://c-swepa.sr.unh.edu/icmecatalogatmercury.html

⁹ https://gong.nso.edu/data/magmap/index.html



Fig. 2. In situ SEP time profiles and plasma and magnetic field observations by STEREO-B, ACE/SOHO, and STEREO-A. Top: Energetic electron and proton temporal profiles observed by, from left to right, STEREO-B, ACE/SOHO, and STEREO-A from comparable energy channels. The CME eruption time is represented by the arrow on the lower x-axis, while the horizontal lines on the upper x-axis of electron panels represent periods of proton contaminating the measured electron fluxes observed by both STEREO. A stream interface observed by STEREO-B is shown as a dash-dotted line, while the salmon shaded areas in STEREO-B and L1 indicate SIRs. The vertical solid lines and grey shaded areas indicate the IP shocks and ICME transits, respectively, associated with the SEP event, based on STEREO ICME list⁴. The IP shocks and ICMEs observed near the Earth, not related with the SEP event, are indicated with vertical dashed lines and green shaded areas, and are based on the SOHO CELIAS/MTOF Proton Monitor shock list⁶ and the near-Earth ICME list⁵, respectively. Bottom: In situ plasma and magnetic field observations by STEREO-B (left), ACE (center), and STEREO-A (right). The panels present, from top to bottom, the magnetic field magnitude, the magnetic field azimuthal and latitudinal angles, ϕ_{B-RTN} and θ_{B-RTN} , the solar wind speed, and the proton density, where RTN stands for radial-tangential-normal coordinates (e.g. Hapgood 1992). The coloured lines in the ϕ_{B-RTN} angle panel indicate the in situ magnetic field polarity, estimated from the magnetic field azimuth. Red and green colours denote inward and outward polarity, respectively. The lower band shows the observed magnetic field polarity, where the yellow intervals represent the periods with the arrow in the solar wind proton panel (IP structures as described in top panels).

corresponding particle energies can be found in Zeitlin et al. (2010). The configuration of spacecraft locations in the heliocentric Earth equatorial (HEEQ) coordinates (e.g. Thompson 2006) early on 2013 August 20, and the in situ measured fluxes of energetic particles from late on 2013 August 19 to August 22 at the different spacecraft locations are shown in Fig. 1. Figure 1(a) shows that STEREO-A, STEREO-B, and L1 are located at he-

liocentric distances near 1 au, while MESSENGER is situated at 0.33 au, and Mars Odyssey is located at 1.58 au. The parent CME apex direction (orange line in Fig.1(a)) is located W171° as seen from Earth, W12° as seen from MESSENGER, W27° as seen from STEREO-A, W43° as seen from Mars Odyssey, and E51° as seen from STEREO-B. The nominal Parker spiral magnetic field lines connecting the spacecraft to the Sun show



Fig. 3. Anisotropy and intensity time profiles of near-relativistic electrons as observed by, from left to right, STEREO-B/SEPT, Wind/3DP, and STEREO-A/SEPT. The first row of panels shows pitch-angle dependent intensity distributions, according to the colour intensity code shown on the upper horizontal bar. The second row of panels indicates the pitch angles covered by the centre of the four telescopes of STEREO/SEPT (sun in red, anti-sun in orange, north in blue, and south in green), and each of the eight pitch-angle bins of the Wind/3DP. The third row of panels indicates the electron flux intensities measured by each of the telescopes or bins, where the vertical arrows indicate the CME onset time. The fourth row of panels represents the first-order anisotropy values, in the range [-3, 3] (e.g. Dresing et al. 2014), but zoomed in on the [-1, 1] interval. The energies examined are 85-125 keV for STEREO and 82-135 keV for Wind.

that MESSENGER, STEREO-A, and Mars Odyssey (separated by 31°) are relatively well connected to the source; and we expected to see similar flux increase behaviours, but this did not occur (Fig. 1(b) to (d)).

To characterize the SEP event at different radial and longitudinal locations, in this section we analyse further the solar energetic particles fluxes and anisotropies at 1 au (determining a reliable first-order anisotropy for MESSENGER at 0.33 au is not possible, due to the lack of sunward-pointing field of view) and the particle timing and magnetic connectivity for different spacecraft. We also examine the particle spectra at STEREO.

3.1. Magnetic connectivity

A fundamental parameter for interpreting the SEP event profiles at different locations is the longitudinal separation between the solar source and the footpoint of the IMF lines connecting to the respective observer. The location and magnetic connectivity around the estimated SEP onset time of the different spacecraft is shown in Fig. 1(a) and detailed in Table 1. Columns (2) to (4), present the spacecraft locations¹⁰¹¹ and, Col. (5) shows the solar wind speed derived from in situ measurements. The nominal magnetic field line length, calculated from $\sim 6 R_{\odot}$ (as the probable particle release height discussed in Sect. 3.5) to the respective spacecraft locations, is shown in Col. (6). Columns (7) and (8) present the footpoint location at 6 R_{\odot} , given by the nominal Parker magnetic field lines connecting the respective spacecraft. The connection angle $(\Delta \phi)$, which is the longitudinal separation between the CME apex direction and the footpoint of the magnetic field line connecting to the corresponding spacecraft, based on the Parker spiral, is shown in Col. (11). MESSENGER, with

Article number, page 6 of 23

 $\Delta\phi$ = +6° and STEREO-A, with $\Delta\phi$ = +29°, are both nominally well connected to the source; the positive sign means westward location with respect to the CME apex direction. This is not the case for STEREO-B and the Earth, which have $\Delta\phi$ over 100°.

The synoptic map from GONG on August 19 at 23:14 universal time $(UT)^{12}$ (not shown) indicates that the STEREO-A and MESSENGER footpoints are situated close to each other and close to the CME eruption location in heliolongitude, but STEREO-A might be separated from the parent active region (AR) by two Heliospheric Current Sheet (HCS) crossings.

3.2. Energetic particle fluxes at 1 au

The upper part of Fig. 2 shows an overview of the energetic particle fluxes observed by STEREO-A (right panels), STEREO-B (left panels), and ACE/SOHO (central panels), from late on 2013 August 19 to August 24. The upper and lower panels display the flux of electrons and protons at different energies: onehour averaged for EPAM and EPHIN, ten-minute averaged for ERNE, SEPT-sun telescope and LET-summed¹³, and fifteenminute averaged for HET. All three spacecraft show a gradual increase in electrons and protons, but the particle intensities observed by the L1 spacecraft show an even more gradual increase than STEREO-B. STEREO-A presents higher intensities than STEREO-B for all comparable energies and species, while SOHO shows intensities similar to those of STEREO-B for electrons in the range of 0.7-3.0 MeV, but lower intensities for protons. However, if cross-calibration is taken into account (Fig. 2 in Lario et al. 2013 and Fig. 3 in Richardson et al. 2014), STEREO-B registers higher intensities than SOHO for all

¹⁰ https://stereo-ssc.nascom.nasa.gov/cgi-bin/make_ where_gif

¹¹ https://omniweb.gsfc.nasa.gov/coho/helios/heli.html

¹² ftp://gong2.nso.edu/oQR/zq5/201308/mrzq5130819/

mrzq5130819t2314c2140_097.gif

¹³ http://www.srl.caltech.edu/STEREO/docs/LET_Level1. html



Fig. 4. Anisotropy and intensity time profiles of protons as observed by, from left to right, STEREO-B/LET, Wind/3DP, and STEREO-A/LET. The panels show almost the same information as in Fig. 3, but with 16 sectors in STEREO/LET, eight front-side (reddish colours) and eight back-side sectors (bluish colours). The energies examined are 1.8-3.6 MeV for both STEREO and 2.1 MeV for Wind. The vertical lines indicate the CME-driven shock arrival to the spacecraft.

comparable energies and species. STEREO electron fluxes below 500 keV (pink line) are susceptible to proton contamination during periods of enhanced sub-MeV proton flux, for example the periods indicated with a horizontal line in both STEREO. From the middle of August 21 to the end of August 23, protons of 500 keV reach the STEREO-A spacecraft and contaminate the electron channels. However, the first part of the event, where the main increase is seen, is clean because electrons arrive much more quickly than the slower low-energy protons causing the contamination. At STEREO-B, the time with proton contamination is observed after the arrival of an IP shock, shown as a vertical solid line.

3.3. Energetic particle anisotropies at 1 au

The directional information provided by different sectors in STEREO/SEPT and LET, and in Wind/3DP, can be used to obtain the electron and proton pitch-angle distribution, where the pitch angles are defined relative to the locally measured magnetic field direction. We analysed the pitch-angle information available from STEREO and Wind, from 2013 August 19 to August 25.

Figure 3 shows electron anisotropies observed by STEREO-A (right panel), Wind (middle panel), and STEREO-B (left panel) around the SEP onset time (from midday of 2013 August 19 to the end of August 20) at energies of 85 to 125 keV for STEREO, and 82 to 135 keV for Wind. We note that the fourth row of panels represents the first-order anisotropy values, in the range [-3, 3] (e.g. Dresing et al. 2014), but zoomed in on the [-1, 1] interval. High positive (negative) values mean that the pitch-angle distribution reaches a clear maximum in intensity at pitch angle 0° (180°), where an anisotropy value close to 0 means isotropic behaviour. In spite of its relatively small connection angle ($\Delta \phi = +29^\circ$), STEREO-A (right panels) observes limited anisotropy during the early phase of the event (although the pitch-angle coverage is not optimal), when the largest anisotropies are usually observed in well-connected SEP events

(e.g. He et al. 2011; Dresing et al. 2014). The small anisotropy observed after the CME eruption, indicated by the arrow, is due to an increase in electrons at the pitch-angle sector closer to 180° (red line in the second panel), which corresponds to particles propagating anti-sunwards, given the negative magnetic polarity. STEREO-B (left panels) does not show clear anisotropies around the onset of the particles either, which is not surprising given its large connection angle ($\Delta \phi = +125^{\circ}$). However, after the onset phase, at the time of the stream interaction region (SIR) observed by the spacecraft (corresponding to the salmon shaded area in Fig. 2), STEREO-B observes a slight increase in electrons near pitch angle 0°. This increase corresponds to particles propagating anti-sunwards due to the positive magnetic field polarity. The middle panels of Fig. 3 show Wind/3DP electron anisotropies, which present a better pitch-angle coverage with eight sectors. The fourth panel shows a period with electron anisotropy that starts around ~03:15 UT on August 20, around five hours later than the CME eruption associated with the main SEP event that originated on the far side of the Sun. This timing agrees with the brightening seen at ~03:20 UT, on the west limb from Earth's perspective (AR3 in Table 3), erupting well after the L1 relativistic 0.25-0.70 MeV electron onset (~01:00 UT). Thus, by reason of timing and magnetic connectivity, this anisotropic period is likely caused by this well-connected small event that overlaps with the main SEP event, and injects electrons in Wind spacecraft. However, during the particle onset and later in the rising phase (i.e. August 20 to August 23) Wind 82-135 keV electrons are contaminated by ions, as observed when compared with the 553 keV proton intensities (not shown), so the data are not reliable.

Figure 4 shows two-minute averaged 1.8 to 3.6 MeV proton anisotropies for STEREO-A (right) and STEREO-B (left), and 2.1 MeV proton data for Wind (middle), from 2013 August 19 to August 25. The fourth row indicates that both STEREO-A (right panel) and STEREO-B (left panel) do not see strong anisotropies. However, from August 21 at 07:00 UT to August 22 at 12:00 UT there is a data interval at STEREO-A

with reduced pitch-angle coverage that leads to a more uncertain anisotropy analysis. At the beginning of the event, after the CME onset indicated by the arrow in the third panel, STEREO-A shows a small proton anisotropy that it is not present in STEREO-B. STEREO-A presents a negative anisotropy most of the time (e.g. the period from mid August 20 to mid August 21), meaning that the particles arrive at pitch angle 180°, which is mainly covered by the front-side sectors (reddish colour), corresponding to particles propagating anti-sunwards. However, during a short period around August 20 17:00 UT, this pitch angle is covered by the bluish colours, which corresponds to particles arriving from the back. This behaviour can be due, for example, to an s-shaped magnetic field in the solar wind. STEREO-B observes a weak anisotropy period starting early on August 21. Later on it shows a sudden increase in the intensity profile associated with a shock arrival on August 22 at 02:09 UT. Around this time we can see how the anisotropy shifts from a positive to negative value due to a change in magnetic field polarity, as the particles are still arriving at pitch angle 0°. The particle arrival direction changes from front-side sectors to rear-side sectors, which suggests that particles observed following the shock are arriving. STEREO-A has a more pronounced intensity increase associated with the shock arrival (third panel on the right) than STEREO-B. When the CME-driven shock arriving to STEREO-A and STEREO-B is the same, this behaviour could be explained by different parts of the shock having different acceleration efficiencies. Wind proton anisotropies (middle panels) are very low during the SEP onset time, due to the very large longitudinal separation between the CME eruption and the field line footpoint of the L1 spacecraft ($\Delta \phi = -117^{\circ}$). On August 21 there are three main spikes present at pitch angle 0° that might not be associated with the main SEP event. As shown in Fig. 2, an ICME and corresponding IP shock, and an SIR arrive at the Earth that day. These anisotropies might be related with in situ particle acceleration or might be SIR-associated.



Fig. 5. Electron peak intensity spectra observed by STEREO-B (left) and STEREO-A (right). The legend shows the fit values: the spectral index below (δ_1) and above (δ_2) the spectral transition; E_b (vertical dashed line); and α , which determines the 'sharpness' of the break (Strauss et al. 2020).

3.4. Electron spectra at STEREO

Following the method described by Dresing et al. (2020) and Strauss et al. (2020), we determined the electron peak intensity spectra, as observed by STEREO/SEPT. Figure 5 presents the fitting, which shows double power-law shapes. The spectral index below and above the spectral transition is respectively $\delta_1 = -1.94 \pm 0.22$ and $\delta_2 = -2.74 \pm 0.14$ for STEREO-A, and $\delta_1 = -1.86 \pm 0.30$ and $\delta_2 = -3.15 \pm 0.33$ for STEREO-B, where the spectral break or transition energy is $E_b = 125 \pm 22$ keV for

STEREO-A and $E_b = 141 \pm 27$ keV for STEREO-B. In comparison to the average values $\delta_1 = -2.53$ and $\delta_2 = -3.93$ of the large statistical sample studied by Dresing et al. (2020), who investigated the electron events observed by STEREO from 2007 to 2018, the spectral indices observed in this event are clearly harder at both STEREO spacecraft. According to Strauss et al. (2020), the observed spectral transitions are likely caused by pitch-angle scattering during transport through the interplanetary medium.



Fig. 6. Velocity dispersion analysis of the onset of the SEP event at STEREO-A. The horizontal and vertical axes correspond to the reciprocal of the particle velocities and onset times, respectively. The red points and circles respectively identify the electron and proton onsets at the corresponding velocities (energies), with the respective errors indicated. The black line is the linear regression fit to all points. The legend gives the effective path length (L) and the estimated release time (onset time) discussed in the text.

3.5. Energetic particle timing

The velocity dispersion analysis (VDA) of an SEP event is based on determining the onset times of the event at a number of energies, and plotting these onset times as a function of the reciprocal of the particle velocities at respective energies. The slope indicates the effective path length (L) and the intercept gives the release time, assumed to be the same for all particles (e.g. Vainio et al. 2013). It is known that the VDA is based on the assumption of a simultaneous release of particles of different energies at the Sun, a scatter-free propagation of the first arriving particles, and an accurate determination of onset times for the in situ observations (Zhao et al. 2019). In this event only STEREO-A shows clear energy dispersion to perform VDA analysis (lower right panel in the upper part of Fig. 2). In the case of STEREO-B (lower left panel in the upper part of Fig. 2), as the intensity enhancement is very gradual, the onsets at different energies can only be calculated using 60-minute averaging, which makes the VDA analysis too ambiguous. For STEREO-A the onset time at different energies from SEPT (65-105 keV electrons) and HET (1.4-2.8 MeV electrons, and 13.6-15.1 MeV, 26.3-29.7 MeV, 60-100 MeV protons) were visually determined, zooming in around the onset time. Low-energy protons from the LET instrument (1.8-3.6 MeV, 4-6 MeV, 6-10 MeV protons) were not included in the VDA analysis as a preceding SEP background seems to mask the true onsets. The linear fit, shown in Fig. 6, derives a VDA effective propagation path length of 3.55 ± 0.50 au,

Table 2. Energetic	particle timing results	and shock height at the	particle release times
<u> </u>		0	1

Spacecraft/Instrument/	Effective path length	Estimated	Estimated Shock heigh		$\theta_{\rm Bn}$ at the	
Particle Species	or IMF length ^a	solar release time	from Sun	centre	cobpoint	
	(au)	(UT) ^b	$(\mathbf{R}_{\odot})^{\mathrm{c}}$		(deg) ^d	
			Parker	ENLIL	Parker	ENLIL
(1)	(2)	(3)	(4)	(5)	(6)	(7)
STA/SEPT/HET ^e	3.55 ± 0.50 (VDA)	23:28 ± 15 min	6.0 ± 2.0	5.7 ± 2.0	22 ± 2	26 ± 3
STA/SEPT/ 65-105 keV e	1.14 ± 0.03 (TSA)	$00:15 \pm 10 \min$	9.5 ± 1.0	10.2 ± 1.0	19 ± 2	20 ± 2
STB/HET/0.7-1.4 MeV e	1.26 ± 0.03 (TSA)	02:57 ± 15 min	$(27.5 \pm 2.0)^{\rm f}$	-	-	-
MESS/EPPS/71-112 keV e	0.31 ± 0.01 (TSA)	$23:22 \pm 5 \min^{g}$	5.9 ± 0.6	5.1 ± 0.6	5 ± 1	8 ± 1
SOHO/EPHIN/0.25-0.70 MeV e	1.09 ± 0.03 (TSA)	$00:58 \pm 60 \text{ min}$	$(15.6 \pm 6.0)^{\rm f}$	-	-	-

Notes.

^a Magnetic field lines length for TSA analysis from ENLIL modelling.

^b Solar release times late on 2013 August 19 and early on August 20, shifted by light propagation time to 1 au in order to compare with electromagnetic observations.

^c CME-driven shock heliospheric height at the cobpoint (Sanahuja et al. 1995) for STEREO-A and MESSENGER (calculated for Parker spiral and ENLIL), and at the shock nose for STEREO-B and SOHO. The range of heights includes the uncertainty in the estimated SEP release time, as discussed in Lario et al. (2016).

 d θ_{Bn} : The angle between the shock normal vector and the magnetic field lines at the cobpoint (estimated using Parker spiral and ENLIL, which are radially extended from 21.5 R_o). The given uncertainties are rounded to one degree. They are based on the shock height deviation due to the uncertainty in the particle release time (the fluctuations in the magnetic field lines, due to waves and turbulence, or in the shock front geometry are not considered).

^e Energies used in VDA analysis: 0.065-2.8 MeV e & 13.6-60 MeV p.

^f STEREO-B is not connected to the shock until later in time, and SOHO is never connected to the shock.

^g MESSENGER was pointing in the anti-Sun direction, thus the solar release time should be interpreted as an upper limit.

much longer than the length of ~ 1.2 au expected for a nominal Parker spiral field and scatter-free propagation. It might indicate either a non-standard interplanetary magnetic field topology or that significant scattering effects are present in the propagation of the particles. It might also be possible that, in this event, the particles of different energies are not simultaneously released from the Sun. The time shifting analysis (TSA), which can be used for the first arriving particles, was also calculated based on Vainio et al. (2013). We visually estimated the onset times of the near-relativistic electron channels for STEREO-A and MESSENGER, the 0.25-0.70 MeV electron channel for SOHO, and the 0.7-1.4 MeV electron channel for STEREO-B. Then we performed a back-shifting of the onset times to get the release times of the electrons at the Sun (shifted by light propagation to 1 au for comparison with electromagnetic observations). Table 2 summarizes the VDA and TSA results for different particle species and spacecraft. Column (2) presents the effective propagation path length for VDA and the ENLIL magnetic field line lengths used in TSA; there would be a difference of less than one minute for all spacecraft if Parker spiral lengths were used. Column (3) shows the estimated particle release time based on VDA and TSA analysis for STEREO-A, and on TSA analysis for STEREO-B, MESSENGER, and SOHO. We note that the MESSENGER onset time should be interpreted as the upper limit since the instrument field of view was pointing to the antisunward direction, and the first particles detected by the antisunward-looking instrument must have been scattered back towards the Sun from beyond the spacecraft location. Column (4) presents the shock heliospheric height at the release time using the coronal shock reconstruction, calculated at the connectingto-observer-point (cobpoint, Sanahuja et al. 1995) for STEREO- A and MESSENGER, and at the shock nose for STEREO-B and SOHO, which is respectively not yet and never connected to the shock (Sect. 4.5). Column (6) presents the angle θ_{Bn} between the shock normal and the Parker field lines at the cobpoint.

Thus, the SEP estimated solar release time, shifted by light propagation time to 1 au, is ~23:22 UT. As explained above, we note that this time should only be considered as an upper limit. At this time the heliocentric height of the quasi-parallel coronal shock is ~6 R_{\odot} , based on the Parker spiral field line.

4. SEP parent solar source: Remote-sensing observations and data analysis

In order to identify the common solar origin for the particle enhancement observed at different spacecraft locations, the time of relevant remote-sensing observations ranges from 2013 August 19 at 20:30 UT to August 20 at 04:00 UT. This interval encompasses about three hours before the first electron onset observed at MESSENGER and the last proton onset at L1. A compilation of videos of the remote-sensing observations referenced in this section are available in the supplementary material of the online version of the paper.

4.1. EUV and coronagraph observations

The EUVI and AIA instruments captured low coronal signatures of CMEs in addition to other forms of solar activity, such as flares and jets. In these images we identified six main ARs as shown in Table 3. Columns (1) to (4) show the AR number (from

A&A proofs: manuscript no. output

AR	Location*	Type of	Time ***	GOES	STEREO-A [#]	STEREO-B [#]	Wind [#]	BIRS [#]
		Activity**		XRS	WAVES	WAVES	WAVES	
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
AR1	E53S18	Brightening	20/08 00:42	B7.5	-	-	-	-
	(11827)							
AR2	W66S04	Brightening	19/08 21:58	C1.0	-	-	-	-
	(11818)	Jet	20/08 00:10	B8.5	Type III (b ⁺)	Type III (b ⁺) (oc.)	Type III (b ⁺)	Type III
		Brightening	20/08 01:40	-	Type III (c ⁺)	-	Type III (c ⁺)	-
		Brightening	20/08 02:44	-	Type III (d ⁺)	Type III (d ⁺) (oc.)	Type III (d ⁺)	Type III
AR3	W67S11	Brightening	20/08 03:20	B6.0	Type III (e ⁺)	-	Type III (e ⁺)	-
	(11817/9)							
AR4	W79S16	Brightening	20/08 01:25	C1.8	-	-	-	-
	(11819)							
AR5	W91S19	Jet	19/08 22:55	B7.0	Type III (a ⁺)	Type III (a ⁺) (oc.)	Type III (a ⁺)	Type III
	(11814)							
AR6 [§]	W171N08	1 st stage	19/08 21:20	-	-	-	-	-
	(11809)	2 nd stage	19/08 22:10	-	-	-	-	-
			19/08 22:28	-	Type II	Type II	-	-
					SA type III $(1->3^+)$	SA type III (1->3 ⁺)	-	-
		M flare starts	19/08 23:30	-	-	-	-	-

Table 3. Active regions present in the period of study and signatures observed by the different instruments

Notes.

(*) Stonyhurst reference frame (AR number from National Oceanic and Atmospheric Administration).

(**) Activity observed by the STEREO/EUVI and SDO/AIA instruments.

(***) UT in year 2013.

(#) Frequency range of 16 MHz to 2.5 kHz for S/WAVES, from 14 MHz to 2.5 kHz for Wind/WAVES, and 14 to 47 MHz for BIRS.

(+) Letters a to e and numbers 1 to 3, as indicated in Fig. 9.

(oc.) Occultation due to relative position between spacecraft and AR.

(§) The AR6 location (represented by the CME apex direction), the CME eruption time, and flare size from this study.

(SA) Shock-accelerated type III radio burst (details given in main text).

east to west longitudes), the AR location, the type of the activity, and the observation time, respectively.

We associate the SEP event with a shock driven by a CME ejecting from AR6, located near the far-side central meridian from Earth's perspective (last row in Table 3), and observed as a two-stage eruption. The CME starting time given by the CDAW SOHO LASCO CME catalogue, calculated at 1 R_{\odot} and referring to the second stage of the eruption, is 22:39 UT; the first C2 appearance time is at 23:12 UT (Fig. 7(e)). The CME is described in the catalogue as a halo-type, with an inferred CME speed of 877 km s $^{-1}$, and a positive acceleration of 4 km s $^{-2}$. The given CME mass and energy values are 1.6×10^{13} kg and 6.1×10^{25} J, respectively, but there is uncertainty due to a poor mass estimate as the mass measurement assumes that the CME material is in the sky-plane and the kinetic energy is obtained from the mass and linear speed (Vourlidas et al. 2000). The CME mass and energy values are relatively large compared to overall CME averaged quantities (Table 1 in Vourlidas et al. 2011). Comparing the dynamical properties of CMEs and SEP intensities from statistical studies (Table 1 in Kahler & Vourlidas 2013), the >50

MeV proton intensity enhancement of this SEP event agrees with the general trend inferred, where higher masses and kinetic energies are associated with higher SEP intensities.

The first stage of the eruption is observed in EUVI-A 171 Å base difference image at ~20:35 UT as very faint and narrow moving loops (online material)¹⁴ that later extend in COR1-A and C2 fields of view, where it is first detected at ~21:20 UT and ~22:12 UT, respectively. Later on, the second stage structures become too dominant to clearly observe the first stage structures. The second stage resulted from a gradual rising motion of a closed coronal structure starting at ~21:50 UT, more clearly seen in difference images, over the west limb from EUVI-A view. This motion suddenly accelerates at ~22:10 UT, making the closed-loop structures move much more quickly. The structures rise over the STEREO-A north-west limb, seen in EUVI-A 195 Å and 171 Å running difference images, as moving loops continuing into COR1-A (online material)¹⁵. The first stage structures

¹⁴ first_stage_lasco_c2_bd.mp4; first_stage_euvia_195_rd.mp4

¹⁵ first_and_second_stage_euvia_195_rd.mp4;

first_and_second_stage_lasco_c2_bd.mp4



Fig. 7. EUV and coronagraph observations by STEREO-B (left), near-Earth (middle), and STEREO-A (right). The upper panels show EUV images taken by EUVI 195 Å and AIA 171 Å (middle). AR6 is indicated by the orange circle in STEREO-A and STEREO-B, and AR2 and AR3 by the purple circle in SDO. The second and third rows show the evolution of the CME at two different times, seen from three different points of view, as observed by the LASCO (magnified image) and SECCHI coronagraphs. The CME-driven shock front is shown with yellow dashed curves (as a rough outline to guide the eye) and the flux-rope structure is indicated with yellow and white arrows. The red arrow indicates the disturbance caused by the first stage (details given in main text). Credit: *JHelioviewer/cdaw.gsfc.nasa.gov*

ture effects can be seen as the distortion of the front, where the north part of the second and main stage is moving more slowly, as it is picking up the first flux-rope, as shown in Fig. 7(d), indicated with a red arrow. Afterwards, this interaction can also be observed as a brighter north part of the CME, indicating increased density, as seen in Fig. 7(g), 7(h), and 7(i), indicated with white arrows. Thus, the interaction of the two stages forms the CME, whose evolution is observed by COR1-B and COR2-B, C2 and C3, and COR1-A and COR2-A, as shown in the second

and third rows of Fig. 7. During this early phase the CME appears as a flux-rope seen face-on by STEREO-B and edge-on by STEREO-A (outlined with the white and yellow arrows in Fig. 7(d) and 7(g), and in Fig. 7(f) and 7(i), respectively). This configuration is in agreement with the orientation of the posteruptive arcades observed after the CME, shown as the orange line in Fig. 7(a) and Fig. 7(c). They show EUVI-B&A 195 Å images, respectively, a few hours after the CME eruption, where the AR6 is indicated with an orange circle. Although the eruption



Fig. 8. EUVI-A observations and analysis. Top: STEREO-A light curve produced using full-disc EUVI 195 Å images, where the horizontal line marks the mean intensity at 01:00 UT. Bottom: Snapshot at 01:00 UT from an EUVI-A 195 Å percentage base-difference movie, where a brightening can be seen in AR6, near the CME dimming area. The yellow circle corresponds to 1 R_{\odot} .

occurs on the far-side hemisphere from the Earth's perspective, SDO/AIA also observes the CME over the north and west limbs between \sim 22:23 UT and \sim 23:10 UT (online material)¹⁶.

The evolution of the CME-driven shock can also be followed by COR1 and COR2, where the shock front is visible from ~22:35 UT onwards. However, it is only at ~22:55 UT when the CME-driven shock can be clearly observed for the first time from the three points of view, COR1-A, COR1-B, and C2 (online material)¹⁷. The yellow dashed curves in the second and third rows of Fig. 7 show the extent of the shock front later in time (as a rough outline to guide the eye, so it does not represent any fitting results). Thus, the global geometry of this CME, formed by two stages, evolves as a single CME driving a single shock.

Weak large-scale disturbances are seen to propagate across the disc in EUVI-A images during $\sim 22:30-22:50$ UT (online material)¹⁸. However, they appear to be qualitatively different from what we know as EUV waves with a continuous welldefined semicircular front (e.g. Thompson et al. 1998).

We produced EUVI light curves for both STEREO spacecraft, only shown for STEREO-A in the upper panel of Fig. 8,

¹⁷ shock_evolution_cor1a_raw_euvi195_diff.mov;

first_and_second_stage_lasco_c2_bd.mp4

which was computed using full-disc EUVI 195 Å images. The small increase in intensity observed from ~22:00 UT to ~22:20 UT is mainly associated with the activity in the west limb from Earth's perspective (purple circle in Fig. 7 (b)), corroborated by the light curve computed only for AR2/AR3 (not shown). From ~22:20 UT to ~23:40 UT the upper panel of Fig. 8 shows the dimming phase, observed as a decrease in brightness that follows the lifting of the CME. Before the dimming stage is finished, a flare starting at ~23:30 UT causes an increase in the intensity of photons/s, peaking at 01:00 UT. The lower panel shows a snapshot from an EUVI-A 195 Å percentage base-difference movie, at the time of the flare intensity peak, where the time of the initial image used is 19:00 UT on 2013 August 19. Thus, the EUVI light curves suggest that the flare represented a post-eruptive arcade (thermal phase) and lacked a clear impulsive phase.

4.2. X-ray observations

Column (3) in Table 3 summarizes the type of solar activity (brightening or jet) observed during the period of study, and Col. (5) indicates the flare size if it was detected by X-ray Sensor (XRS) on board GOES¹⁹. XRS observes the Sun continuously from Earth's vantage point in two broadband soft X-ray channels. For the purposes of detecting the onset and the intensity of solar flares, it is usually sufficient to monitor the wavelength 1-8 Å band. The lack of intense emissions observed during the period of study stems from the activity on the far side of the Sun from Earth's perspective. However, the GOES soft X-ray flux equivalent level of the activity observed on the far side, can be derived using the EUVI/195 Å light curve as they are dominated by the Fe XXIV line at 192 Å during flare times (Nitta et al. 2013). This process yields an equivalent GOES soft X-ray level of 3.3×10^{-5} W m⁻² for STEREO-A and 3.1×10^{-5} W m⁻² for STEREO-B. Due to the uncertainties involved in the process, the corresponding GOES soft X-ray level is calculated with an error of a factor of three (scattering present in Fig. 7 in Nitta et al. 2013). Therefore, the equivalent GOES soft X-ray is within the range from M1 to M9, generally an M-class flare.

4.3. Radio observations

Radio observations provide insight in the coronal and IP processes that might be present related with the particle acceleration. Columns (6) to (9) in Table 3 present a summary of the different radio signatures observed by the S/WAVES, Wind/WAVES, and BIRS instruments during the period of study. The high- and low-frequency radio emission observed by the S/WAVES instruments (from 16 MHz to 2.5 kHz) on STEREO A and B and the Wind/WAVES experiment (from 14 MHz to 2.5 kHz) are shown in Fig. 9. Both S/WAVES present similar and complex radio emission, with intense activity during the period of observation. A first type II radio-emission is seen at ~22:28 UT at 16 MHz (number 1 in Fig. 9), drifting slowly to 300 kHz at ~04:00 UT. The type II emission could have started earlier, but there is no information above 16 MHz. These type II bursts, due to their low-frequency emission (Pohjolainen et al. 2007, and references therein), might be identified as coronal-IP type II radio bursts, associated with the shock driven by the CME, observed in COR1 at ~22:35 UT. The IP type II emission persists for ~ 8 hours (it can be tracked up to 06:00 UT, not shown). This time is longer that the average duration of \sim 3-4 hours derived

¹⁶ second_stage_observed_by_sdo_aia193_pbd.mov

shock_evolution_cor1b_raw_euvi195_diff.mov;

¹⁸ large_scale_weak_disturbances_euvia195_pbd.mov

¹⁹ ftp://ftp.swpc.noaa.gov/pub/warehouse/2013/2013_
plots/xray/



Fig. 9. Radio dynamic spectra observed by both S/WAVES and Wind/WAVES from 2013 August 19 at 21:00 UT to 2013 August 20 at 04:00 UT. Both S/WAVES present coronal–IP type II radio emission, with fp and 2fp clearly present from ~22:28 to ~04:00 UT. Groups of (probably SA) type III bursts can be seen emerging from the type II, from ~22:28 UT (labelled with an orange '1') to ~00:00 UT (orange '3'). Both type II and type III bursts are time related with the CME ejecting from AR6 (details given in main text). The orange number 2 indicates where fp and 2fp could possibly be further broken down into more bands. S/WAVES-A and Wind/WAVES present flare-related type III bursts at different times (green letters a to e), caused by the activity in AR2, AR3, and AR5, and S/WAVES-B observes some of these flare-related type III bursts lower in frequency, due to occultation (Table 3). Credit: *Observatoire de Paris-Meudon*



Fig. 10. Height evolution of the CME bulk and LE, coronal shock, and type II radio burst. The brown, green, and red lines show the CME (bulk and LE) and coronal shock height fits, respectively, from GCS and spheroid reconstruction. Type II radio burst drift, using the hybrid density model, is represented by the purple line (details given in main text). The legend shows the speeds corresponding to each linear fit.

from the >50 MeV SEP statistical study performed by Kouloumvakos et al. (2019), where in some extreme events the type II emission lasts for ~12 hours. Assuming that the 16 MHz emission is related with the fundamental (fp), and using the hybrid

density model (Vršnak et al. 2004, details given below), this frequency corresponds to a heliocentric height of ~2.5 R_{\odot} for the burst driver in the corona. A probable second harmonic, 2fp, might be present from ~22:55 UT (at 16 MHz) to ~04:00 UT (at 600 kHz), and at 23:00 UT (at \sim 10 MHz, number 2 in Fig. 9) the fp and 2fp are possibly further broken down into more bands (Pohjolainen et al. 2007). Also at \sim 23:00 UT the type II emission is more prominent starting at ~5 MHz, which corresponds to a heliocentric height of ~4.5 R_{\odot} . The CME-driven shock speed can be estimated analysing the frequency drift. However, the heights and speeds inferred from radio burst dynamic spectra need to be handled with care since they strongly depend on the chosen electron density model. To estimate the CME speed from radio observations, we assumed that the type II burst emission is formed by accelerated electrons, near the shock, at the leadingedge (LE) of the CME, although we can expect a certain offset between the burst driver and the shock (Jebaraj et al. 2020). The selection of a density model includes the knowledge of several conditions (Pohjolainen et al. 2007). In this case, the hybrid model (Vršnak et al. 2004) is chosen because it more closely fits the shock height-time evolution as observed by COR1-A and COR2-A. The hybrid model is a mixture of the Saito et al. (1970) and the Leblanc et al. (1998) models, with some small changes, and it can be used for connecting bursts in the corona and the IP space. Following the hybrid model, and knowing the frequency of the type II emission at a given time, we derived the heliocentric height of the burst driver in solar radii. Figure 10 shows that the frequency drift of the fundamental yields a linear fit speed



Fig. 11. Coronagraph images and GCS 3D reconstruction for the CME (green mesh) and associated driven shock (red mesh) as seen via three different instruments: COR2-B ((a),(e)), C3 ((b),(f)) and COR2-A ((c),(g)). The C3 images are magnified for better visualization. The right plots ((d), (h)) show the 3D representation, in HEEQ reference in R_{\odot} , of the CME and the driven shock, together with the nominal Parker spirals (solid lines) and the lines connecting the Sun (dashed lines), for STEREO-A (STA in red), STEREO-B (STB in blue), MESSENGER (MESS in orange), and the Earth (L1 in green). The Sun is shown as a yellow circle (not to scale).

of 1080 ± 46 km s⁻¹ (purple line), which is the same for both S/WAVES within the given uncertainty. The points were taken from 22:45 UT (8.5 MHz, i.e. 3.2 R_☉) up to 00:00 UT (1 MHz, corresponding to 10.4 R_☉), when the lowest frequency available from the hybrid model was reached. Based on the type II radio burst fit, the derived height of the CME-driven shock at the estimated particle solar release time is 6.6 ± 0.5 R_☉.

Wind/WAVES presents, simultaneously with S/WAVES, flare-related type III bursts from at least 16 MHz to lower frequencies at different times (letters a to e in Fig. 9), as summarized in Table 3, that are not related with the SEP event. In particular, the activity on the west limb from Earth's perspective associated with the anisotropy period observed at Wind (Sect. 3.3) might be related to the type III radio burst labelled 'e'. Some of these type III radio bursts start higher in frequency, as they can be seen in the BIRS experiment, from 14 to 47 MHz (Col. 9 in Table 3), and in Learmonth spectrograph, from 25 to 180 MHz (except type III at 22:55 UT). Both S/WAVES present groups of type III emission bursting for several minutes, and covering 16 MHz to lower than 100 kHz in frequency, from ~22:28 UT to ~00:00 UT (from number 1 to number 3 in Fig. 9). This type III emission is relatively intense at low (hectometric) frequencies, and it is enhanced as observed by S/WAVES-A at around 23:00 UT (number 2 in Fig. 9). It seems that the type III bursts are emerging from the type II emission, but there is no radio information above 16 MHz to confirm it. These type III bursts can be observed down to 300 kHz (first group) and even lower (the other groups), which suggests that electrons are being lost to the IP medium through open magnetic field lines, as normally happens in energetic events (Cane et al. 1981). These type III bursts could be identified as shock-accelerated type III bursts showing particular properties: their timing is not associated with any starting or peaking flare (light curve in Fig. 8), and they seem to be produced from type II bursts. The corresponding SEP event shows proton increases of at least 50 MeV, consistent with the previous studies about the relationship between radio emission and SEP events; for example Cane et al. (2002) concluded that >20 MeV proton increases were always accompanied by type III or type III-1 bursts. The M-class flare seen on AR6, starting at ~23:30 UT and with a brightening peak at ~01:00 UT, is not associated with any type III burst, suggesting that there are no accelerated electron beams escaping from the flare region.

4.4. CME 3D reconstruction

To characterize the CME associated with the SEP event, mainly in terms of CME speed, width, and location, we took advantage of the multi-view spacecraft observations and reconstructed the 3D CME using the GCS model. As described in Thernisien et al. (2006) and Thernisien (2011), the GCS model uses the geometry of what looks like a hollow croissant to fit a flux-rope structure using coronagraph images from multiple viewpoints. The GCS model was developed to integrate both the self-similar expansion and the flux-rope three-dimensional morphology, using the following constraints: the legs of the structure are conical, the front is pseudo-circular, the cross section is circular, and it expands in a self-similar way. GCS estimates the CME position, direction, 3D extent, 3D speed, and the CME orientation. The sensitivity (deviations) in the parameters of the GCS analysis is given in Table 2 of Thernisien et al. (2009). The COR1/2-A and COR1/2-B, and C2 and C3 quasi-simultaneous images were used to fit the flux-rope shape of CME at different times, where COR1 and C2, and COR2 and C3 data were used at lower and higher altitudes, respectively, to help track the propagation and evolution of the CME more precisely. The routine used for the reconstruction is rtcloudwidget.pro, available as part of the scraytrace package in the SolarSoft IDL library²⁰.

²⁰ http://www.lmsal.com/solarsoft/

Figure 11 shows the coronagraph images (upper left) and GCS fit analysis (lower left) for the CME, on August 20 at 00:24 UT, where the green mesh represent the flux-rope structure erupting from AR6, seen from the points of view of STEREO-B (e), SOHO (f), and STEREO-A (g). Figure 11(g) reveals that the CME is more oriented towards STEREO-A than to STEREO-B, as the green wire-frame is more symmetric with respect to the centre than in STEREO-B image (e). Figures 11(d) and 11(h) show the 3D reconstruction of the flux-rope structure at two different times, and its relative position to the Sun-spacecraft direction for STEREO-A, MESSENGER, STEREO-B, and L1.

The CME 3D reconstruction, from ~3.5 to ~21 R_{\odot} , presents a smooth behaviour with minor variations in the values of its longitude (~+171°) and tilt (~36°), which is the angle relative to the solar equator. However, the structure seems to experience a slight southward deflection, from a CME apex of +8° at 23:25 UT, corresponding to ~6 R_{\odot} , to -5° at ~ 00:54 UT, corresponding to ~15 R_{\odot} . The aspect ratio (~0.4) remains almost constant at all times, following the hypothesis of self-similar expansion, and so does the half-angle (~59°).

The linear fit of the height evolution in time for the CME bulk and LE are shown in Fig. 10. The calculated bulk speed (i.e. speed of the centre of the flux-rope at the apex) is 802 \pm 23 km s⁻¹, while the LE speed is 1149 \pm 33 km s⁻¹. The width of the CME was estimated based on Dumbović et al. (2019), where the angular extent in the equatorial plane is represented by $R_{\text{maj}} - (R_{\text{maj}} - R_{\text{min}}) \times |tilt|/90$, obtaining a result of 121°. The value of R_{maj} (face-on CME half-width) was calculated adding R_{min} (edge-on CME half-width) to the half-angle, and R_{min} was calculated as the arcsin(*ratio*). The CME width deviation was derived from the mean half-angle error, estimated by Thernisien et al. (2009) as +13°/-7°. Thus, the CME is ejecting from W171N08 and it is wide (~121°) and fast (~1149 km s⁻¹).

4.5. Coronal shock 3D reconstruction

In order to gain a detailed understanding of the magnetic connectivity to the CME-driven shock associated with the SEP event, the coronal shock 3D reconstruction was performed using the model developed by Olmedo et al. (2013). This model fits the coronal shock to an spheroid shape, although the real wave probably presents some differences from the ideal contour (Susino et al. 2015). Some examples of the application of the model can be found in Mäkelä et al. (2015) and Xie et al. (2017). The model used here for the shock reconstruction is similar to that applied by Kwon et al. (2014), who used an ellipsoid shape for the fitting. As there is no reference in the literature that describes the spheroid model, we include the details of the process here. The cartesian equation for a spheroid that is rotationally symmetric along the z-axis is given by $(x^2 + y^2)/a^2 + z^2/c^2 = 1$, where the two distinct semi-axes are denoted a in the azimuthal direction and c in the radial direction. The geometry of the spheroid is defined by three parameters: the height (h) of the spheroid in the radial direction from the solar centre in units of solar radii (i.e. the height of the nose of the spheroid in the z-axis), the selfsimilarity coefficient (κ), and the ellipticity (e). The semi-axis (a) of the spheroid in the azimuthal direction is related with the parameter κ such that $a = (h - 1) \times \kappa$. For a given height (h), and knowing the self-similarity parameter (κ), which takes values between 0.01 and 2, the semi-axis (a) in the azimuthal direction is determined. The ellipticity (e) is analogous to the eccentricity, and if the ellipticity is known, the radial semi-axis (c) can be calculated.

The model uses a spheroid shape to fit the CME-driven shock using quasi-simultaneous images from COR1 and COR2, and from C2 and C3. The images underwent a basic process for calibration, and base-difference or running-difference procedure was used to highlight the front of the shock better. By varying the ellipticity (*e*), which is always positive, the relationship between *c* and *a* can be changed. For c > a a prolate spheroid is defined, and an oblate spheroid for c < a. In the routine, however, the difference between an oblate or prolate spheroid is implemented by choosing a negative or positive value of the ellipticity (*e*), respectively. To determine the latitude and longitude coordinates of the source region, we calculated the intersection point of the radial axis (z-axis) of the spheroid and the solar surface. The routine used for the reconstruction is also *rtcloudwidget.pro* (reference in Sect. 4.4).

The main shock reconstruction period, using the three vantage points of view, covered from 22:55 UT on August 19 when the shock height was ~3.5 R_o, to 01:54 UT on August 20, corresponding to a shock height of ~21 R_o. Signatures of the shock formation are observed earlier in EUVI and COR1 images, from ~22:25 UT and ~22:35 UT onwards, respectively, when no observations from the Earth's perspective were available. Using these data we also performed an earlier 3D reconstruction of the CME-driven shock. However, as the shock front is not as clear as from ~22:55 UT onwards, we considered that the shock reconstruction using EUV images for the period prior to this time was less reliable. Therefore, we based our further analysis of the 3D shock reconstruction on white-light coronagraph data from three viewpoints.

The CME-driven shock 3D reconstruction at 00:24 UT on August 20 is shown in Fig. 11. The red mesh seen overplotted on the coronagraphs images represents the 2D projections of the 3D reconstruction of the coronal shock, as seen from COR2-B (e), C2 and C3 (f), and COR2-A (g). The 3D view of the CME-driven shock and the nominal Parker magnetic field lines, starting at 2.5 R_o, for STEREO-A, STEREO-B, Earth and MESSENGER, are shown in Fig. 11(d) and 11(h). We note that the rear part of the spheroid was removed, exactly in the plane containing the source origin of the shock at the Sun. It is remarkable that the nose of the CME-driven shock is oriented towards MESSEN-GER and STEREO-A, and the magnetic field lines connecting STEREO-A and MESSENGER intersect the spheroid representing the CME-driven shock.

The 3D reconstructed shock parameters are consistent during the main reconstruction period. The resultant spheroid is oblate (e=0.46) and the self-similarity coefficient (κ) is ~0.69. The longitude and latitude values show that the origin at the Sun of the coronal shock is located at W173N02, slightly west and south from AR6 (W171N08). Lastly, the coronal shock speed, estimated as the linear fit of the evolution of the shock height, is 1160 ± 21 km s⁻¹ (red line in Fig. 10). We note that this shock speed is very similar to the estimated type II drift (Sect. 4.3 and purple line in Fig. 10). Based on the spheroid model analysis, the derived height of the CME-driven shock at the estimated particle solar release time is 6.0 ± 0.5 R_{\odot}.

Table 4 shows the (analytically calculated) intersection parameters between the spheroid and the nominal Parker magnetic field lines connecting to each spacecraft. The first two lines in brackets correspond to the earlier 3D shock reconstruction mentioned above. Column (2) shows the first time the reconstructed CME-driven shock intersects the magnetic field lines connecting the spacecraft. Column (4) represents the coronal shock heliospheric height at the intersection time. Column (6) presents θ_{Bn} , which is the value of the angle between the shock normal

A&A proofs: manuscript no. output

|--|

s/c	Estima	Shock height from		$\theta_{\rm Bn}$ at the		Cobpoint		
	intersection	Sun centre (R_{\odot})		cobpoint (deg) ^c				
	Parker	ENLIL	Parker	ENLIL	Parker	ENLIL	Parker	ENLIL
(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
(STA	22:41 ± 10 min	$22:42 \pm 10 \min$	2.6 ± 0.3	2.8 ± 0.3	37 ± 2	24 ± 2	E157S04	E164S05)
(MESS	22:38 ± 10 min	$22:42 \pm 10 \min$	2.5 ± 0.3	3.0 ± 0.3	5 ± 1	9 ± 1	W179N01	W174N01)
STA	$22:55 \pm 10 \min$	$22:55 \pm 10 \min$	3.3 ± 0.3	3.2 ± 0.3	17 ± 2	22 ± 2	E157S04	E164S05
MESS	$22{:}55\pm10~\text{min}$	$22:55 \pm 10 \min$	3.5 ± 0.3	3.5 ± 0.3	4 ± 1	7 ± 1	W179N01	W174N01

Notes.

^a As the estimated first intersection times occur in the first step of the coronal shock reconstruction using coronagraph data from both STEREO and LASCO (last two lines), we also present the first intersection results calculated using EUVI/COR1 data from both STEREO instruments (first two lines in brackets). Details given in main text.

^b Time on 2013 August 19.

^c Angle between the shock normal and the upstream magnetic field line at the cobpoint. The given uncertainties are rounded to one degree. They are based on the shock height deviation due to the coronagraph images cadence (the fluctuations in the magnetic field lines, due to waves and turbulence, or in the shock front geometry are not considered).

and the upstream magnetic field line at the cobpoint. Column (8) shows the cobpoint coordinates. Thus, the first connection to the shock, for STEREO-A (STA) and MESSENGER (MESS), is at least ~27 minutes before the particle solar release time at $23:22 \pm 8$ min. Taking into account the earlier 3D reconstruction, the first connection to the shock for STEREO-A and MESSEN-GER would be ~22:41 UT and ~22:38 UT, respectively, which is ~44 minutes before the particle solar release time. Based on the shock reconstruction up to ~21 R_{\odot} , STEREO-B and L1 are not connected to the coronal shock at any time. The heliocentric shock height near the shock nose at the first intersection time is ~2.5 R_{\odot} , in comparison with the ~6 R_{\odot} at the estimated particle release time. Both MESSENGER and STEREO-A, connecting to the CME front area, are observing quasi-parallel shocks. The MESSENGER cobpoint is located near the shock nose and the STEREO-A cobpoint is longitudinally separated less than $\sim 30^{\circ}$ from the CME-driven shock nose.

5. Heliospheric conditions: In situ solar wind observations and ENLIL simulation

5.1. In situ plasma and magnetic field observations

The heliospheric conditions at the time of the particle release, in which the particles propagate, can significantly affect the SEP timing and intensity profiles. Multi-point solar wind and IMF observations provide a comprehensive understanding of the geometry, not only of the IP structures and their possible influence in the propagation of the SEPs, but also of the shocks and their role in forming the intensity-time profiles.

As presented below, according to in situ solar wind and magnetic field data, an IP shock and ICME was observed at MES-SENGER, and both STEREO spacecraft, consistent with the two-stage CME erupting from AR6, as no other CMEs within a few hours were ejected from the same AR and surrounding region. The detailed analysis of the IP shock and ICME observed by MESSENGER, at 0.33 au, and by STEREO-A and STEREO-B, at ~1 au, is the focus of a second study (Rodríguez-García et al. 2021). Because the Earth is located on the opposite side of the Sun from AR6, it is exceedingly unlikely that the reported ICME transients from August 22 to August 24 on the near-Earth ICME list by Richardson and Cane are associated with the SEP parent CME.

In situ plasma and magnetic field observations by PLASTIC and the Magnetic Field Experiment on board STEREO, and by SWEPAM, SWICS (proton density), and the Magnetic Field Experiment on board ACE are shown in the lower part of Fig. 2, where the different IP structures observed by the spacecraft were classified using the lists summarized in Sect. 2.

The left panels of bottom part of Fig. 2 show that at the time of the relativistic electron onset (arrow in the speed panel), STEREO-B is embedded in a slow solar wind stream ($V_{sw} \sim 320$ km s⁻¹). A short while later, during the rising phase of the SEP event, STEREO-B observes a SIR, corresponding to the salmon shaded area, where the stream interface is represented by the grey dash-dotted line. The arrival of the IP shock, at 02:10 UT on August 22, is indicated with a vertical black line. STEREO-B, located 53° west of CME apex direction (estimated at the time of the ICME arrival), shows a gradual onset, with increased intensities as the spacecraft becomes connected to the shock. Maximum intensity, for <10 MeV protons, is observed when the spacecraft encounters magnetic field lines that are connected to the shock when it is beyond 1 AU, with an exponential decay following. This peak observed after the shock can be related with the spacecraft connecting to a stronger region of the shock closer to the nose (Cane et al. 1988), although it has little influence on the near-relativistic electrons and protons >30 MeV. Thus, STEREO-B and also STEREO-A (discussed below) match the expected behaviour for a gradual SEP event seen from two different solar longitudes relative to the CME-driven shock (Cane et al. 1988). This behaviour is also consistent with the observed anisotropies (Sect. 3.3). The ICME arriving at STEREO-B is shown as a grey area and seems to have little influence in the SEP profile. The solar wind after the ICME is a fast wind stream with Alfvénic-like fluctuations in the magnetic field, as some correlation is observed between the magnetic field and solar wind speed in the normal (N) component from RTN coordinate system.

The right panels of Fig. 2 show that at the time of the relativistic electron onset, the solar wind speed at STEREO-A is ~400 km s⁻¹. The IP shock arrives at 07:05 UT on August 22. STEREO-A, located 24° east of the CME apex direction (estimated at the time of the ICME arrival), observes a peak in intensity for < 6 MeV ions when the shock arrives, followed by a decay. The local shock has little influence on the near-relativistic electrons (although this is difficult to evaluate, due to proton contamination around the shock transit time), and for > 10 MeV. A sudden decrease in protons up to 30 MeV is observed at the time of the ICME arrival, at 23:15 UT on August 22, followed by a transient depression around 12 hours later. The decrease observed in near-relativistic electrons at the same times is unsure due to proton contamination.

The central panels of Fig. 2 show that at the time of the relativistic electron onset the solar wind speed at L1 is ~450 km s⁻¹. In a later phase of the SEP event, not affecting the onset, the near-Earth spacecraft are observing several IP structures not related with the activity in AR6, located close to the farside central meridian from Earth's perspective. Just before the beginning of August 21, a first IP shock (vertical dashed line), and the corresponding ICME (first green shaded area), embedded in a SIR (salmon shaded area, based on Chi et al. 2018) arrive at the L1 spacecraft. The perturbed IP magnetic field lines may be affecting the SEP time profile, with a possible additional contribution of in situ particle acceleration (protons up to several MeV). In particular, the two small peaks observed at the beginning and ending of August 21 could be related to the local compression, observed as increased magnetic field, density, and speed. This behaviour is also consistent with the observed anisotropies (Sect. 3.3). A second ICME (second dashed line and second green shaded area) arrives before the end of August 22 with little influence on the SEP profile. The last shock arrives on August 24, but no significant effect is observed on the SEP time profile either.

The energetic particle and magnetic field observations by MESSENGER, in orbit around Mercury and located 6° east of the CME apex direction (estimated at the time of the ICME arrival) are shown in Fig. 12. The grey shaded areas delimit the intervals related with Mercury's magnetosphere, identified by eye, while outside these intervals the spacecraft was in the solar wind. We observe a sharp decrease in the SEP profile starting with the arrival of the IP shock transit on August 20 at 12:40 UT (vertical black line). The influence of the ICME (green shaded area) arriving on August 20 at 17:49 UT, with a change in the magnetic field polarity, and ending on August 21 at 08:38 UT, cannot be evaluated due to the lack of particle data during this period.

5.2. ENLIL model

The WSA-ENLIL + Cone model (Odstrcil et al. 2004) is a global 3D MHD model²¹ that provides a time-dependent background characterization of the heliosphere outside of 21.5 R_{\odot}. ENLIL uses time-dependent GONG magnetic field data as a basis, into which spherical shaped high-pressure structures without any internal magnetic field are inserted to mimic observed CME-associated solar wind disturbances. The ENLIL-modelled CME has an artificially higher pressure to compensate for the lack of a strong magnetic field. At the CME nose, the dynamic pressure dominates and the compression is strongest, thus the ENLIL+Cone model can often simulate the CME nose well, de-



Fig. 12. MESSENGER energetic electron and magnetic field observations. The panels show from top to bottom: Near-relativistic electron intensities, magnetic field magnitude, and magnetic field angles $\phi_{\text{B-RTN}}$ and $\theta_{\text{B-RTN}}$. The coloured lines and lower band in the $\phi_{\text{B-RTN}}$ angle panel as in Fig. 2. The grey shaded areas delimit the intervals related with Mercury's magnetosphere. An IP shock and ICME arriving at MES-SENGER are indicated by the black vertical line and the green shaded area, respectively.

spite the lack of internal magnetic field for the CME driver. In contrast, at the flank or edge of a CME, the interaction of CME magnetic field with background solar wind cannot be ignored anymore, so the lack of internal magnetic field would influence the simulation of CME flanks. This affects the shock parameters such as the shock obliquity derived from the model (Bain et al. 2016). The reliability of ensemble CME-arrival predictions depends strongly on the initial CME input parameters, such as speed, direction, and width (Mays et al. 2015; Kay et al. 2020), but also on the errors that can arise in the ambient-model parameters and on the accuracy of the solar-wind background derived from coronal maps. Based on Wold et al. (2018), the mean absolute arrival-time prediction error is 10.4 ± 0.9 hours, with a tendency to an early prediction error of -4.0 hours.

The magnetic connectivity at the onset time can be relevant to the understanding of the SEP observations, and considering the ENLIL-modelled varying solar wind conditions to calculate the IMF lines is an alternative to using the nominal Parker spirals. The preconditioning of the heliosphere and the interaction of the IP structures that might be present at the onset time can actively influence this connectivity. Therefore, the ENLIL simulation time ranges from August 16 to August 26 (i.e. from four days before to six days after the SEP event). This interval encompasses the possible previous CME and CME-driven shocks that may influence the particle propagation at the onset time, and the ICME evolution through the IP medium up to 2.1 au. For this purpose, the GCS 3D reconstruction process presented in Sect. 4.4 was also used for the five relevant prior CMEs erupting in the time range of August 16 to August 20. For this simulation, we used the CME LE parameters (position and speed) rather than the bulk (bright core, if present) as they often capture the overall and great impact of the high-pressure structures better. For the

²¹ https://ccmc.gsfc.nasa.gov/models/modelinfo.php?
model=ENLIL%20with%20Cone%20Model

Table 5. Time line of the 2013 August 19 SEP event.

Da	ate	Delay to	Event	Mission/	This study/	Notes
(U	T)	CME (hr) ^a	description	instrument	reference	
19/08	20:35	-1:35	CME starts (1 _{st} stage)	STA/EUVI	Sect. 4.1	AR6 (W171N08)
19/08	22:10	0:00	CME starts (2 _{nd} stage)	STA/EUVI	Sect. 4.1	AR6 (W171N08)
19/08	22:28	0:18	Type II bursts (fp)	STA & STB/WAVES	1 in Fig. <mark>9</mark>	${\sim}16$ MHz/2.5 R_{\odot}
19/08	22:28	0:18	SA type III bursts	STA & STB/WAVES	1 in Fig. 9	First group
19/08	22:35	0:25	Coronal shock in COR1	STA/SECCHI	Sect. 4.5	W173N02
19/08	22:42	0:32	First shock connection	STA& MESS	Table 4	EUVI/ENLIL
19/08	23:00	0:50	SA type III bursts	STA & STB/WAVES	2 in Fig. 9	Second group
19/08	23:19	1:09	71-112 keV e onset	MESSENGER/EPS	Fig. 1(c)	Anti-Sun /±5 min
19/08	23:22	1:12	71-112 keV e SRT ^b	MESSENGER/EPS	TSA (time shifted)	Anti-Sun /±5 min
19/08	23:30	1:20	~M soft-X ray starts	STA/EUVI	Light curve in Fig. 8	±5 min
20/08	00:15	2:05	0.7-1.4 MeV e onset	STA/HET	Fig. 2 (top right panel)	±15 min
20/08	00:15	2:05	60-100 MeV p onset	STA/HET	Fig. 2 (top right panel)	±15 min
20/08	00:25	2:15	65-105 keV e onset	STA/SEPT	Fig. 2 (top right panel)	±10 min
20/08	01:00	2:50	~M soft-X ray peaks	STA/EUVI	Light curve in Fig. 8	±5 min
20/08	01:00	2:50	0.25-0.70 MeV e onset	SOHO/EPHIN	Fig. 2 (top central panel)	±60 min
20/08	03:00	4:50	0.7-1.4 MeV e onset	STB/HET	Fig. 2 (top left panel)	±60 min
20/08	03:00	4:50	25-50 MeV p onset	SOHO/EPHIN	Fig. 2 (bottom central panel)	±10 min
20/08	04:00	5:50	>27 MeV p onset	MARS ODY./HEND	Fig. 1(b)	±20 min
20/08	06:00	7:50	~1.5 days anisotr. period	STA/LET	Fig. 4 (right panels)	1.8-3.6 MeV p
20/08	06:45	8:35	60-100 MeV p onset	STB/HET	Fig. 2 (bottom left panel)	±15 min
20/08	12:41	14:31	IP shock arrival	MESSENGER	ICME catalogue	975 km/s ^c
20/08	17:49	17:39	ICME starts	MESSENGER	In situ IMF data	860 km/s ^c
21/08	05:00	30:50	~1.5 days anisotr. period	STB/LET	Fig. 4 (right panels)	1.8-3.6 MeV p
21/08	08:38	34:28	ICME ends	MESSENGER	In situ IMF data	-
21/08	22:01	47:51	First shock connection	STEREO-B	ENLIL model	-
22/08	02:09	51:59	IP shock arrival	STB	IP Helsinki DB	615 km/s ^d
22/08	07:05	56:55	IP shock arrival	STA	IP Helsinki DB	483 km/s ^d
22/08	23:15	73:05	ICME starts	STA	ICME catalogue	475 km/s ^d
23/08	07:00	80:50	ICME ends	STB	ICME catalogue	-
24/08	23:25	97:15	ICME ends	STA	ICME catalogue	-

Notes.

^a The second and main stage of CME is taken for time reference.

^b Solar release time (upper limit), shifted 8.3 min for comparison with electromagnetic observations from 1 au.

^c Average transient speed from the Sun to spacecraft location.

^d In situ speed of the transient.

study of this particular scenario, we performed two simulations with two different grid resolutions (low and medium), but otherwise identical runs. The medium-resolution run (786x60x180 for 2.1 au radius, $\pm 60^{\circ}$ latitude, 360° longitude) seems to increase

the difficulty of the model in deriving the ICME flank geometry, observed by STEREO-B. The derived shock parameters from the medium-resolution grid fluctuate significantly, while those from the low-resolution grid (384x30x90 for 2.1 au radius, $\pm 60^{\circ}$ lat-



Fig. 13. Radial velocity contour plots from the ENLIL simulation in the ecliptic plane. The black and white dashed lines represent the IMF lines and the black contours track the ICMEs. The white lines correspond to the HCS, which separates the regions with opposite magnetic polarity, shown in blue (negative) or red (positive) on the outer edge of the simulation region. Left panel (a): Magnetic connectivity of the different spacecraft around the particle solar release time. Right panel (b): Magnetic field line connecting STEREO-B intersects for the first time the IP shock driven by the ICME. Credit: *CCMC*

itude, 360° longitude) match the in situ data more closely. The CME and model set-up parameters, and the results of the simulations are available on the Community Coordinated Modeling Center (CCMC) website²² ²³.

The information derived from the modelled IMF lines at the SEP onset time, using the medium-resolution run, which provides the information about the magnetic field lines with smaller uncertainty, is summarized in Table 1 and represented in the left panel of Figure 13. As the magnetic footpoints and the IMF lines passing through the different spacecraft are given by the model, the length of the magnetic field lines can be calculated. For each spacecraft, Cols. (5), (6), (9), and (10) of Table 1 show, respectively, the solar wind speed, the magnetic field line length, and the longitude and latitude of the footpoint locations on August 19 at 23:00 UT. The ENLIL lines are computed up to 21.5 R_{\odot} , and radially extended to 5 R_{\odot} , which corresponds to the height of the shock at the estimated particle released time (~23:22 UT), based on the MESSENGER TSA analysis presented in Table 2. At this height (5 R_{\odot}), the shock is simulated as a quasi-parallel shock in the corona based on the spheroid model and ENLIL magnetic field lines ($\theta_{Bn} = 8^{\circ}$ in Col. (7) in Table 2).

Based on the connection angle shown in Col. (11), STEREO-B and Earth seems to have respectively a better and worse magnetic connection to the source area (W171N08) from the ENLIL model than they would have from the nominal Parker spiral. Column (6) shows that the ENLIL-modelled and Parker spiral magnetic field lines lengths are comparable for all spacecraft. Figure 13 (a) shows a snapshot of the ENLIL simulation at the onset time, where the black contours track the ICMEs, as they are set as regions when the density ratio to the background is higher than a certain threshold. This figure reveals that there are IP structures present near STEREO-B and Earth that might be mod-

Rodriguez-Garcia_093020_SH_1.php (low resolution)

ifying the connectivity at these locations with respect to nominal Parker spirals. A region with increased solar wind speed might arrive at STEREO-B at around the SEP onset time, which could correspond to the SIR observed in Fig. 2. STEREO-A might be separated from MESSENGER by an HCS crossing.

Table 4 presents the parameters of the intersection between the reconstructed coronal shock and ENLIL magnetic field lines (extending radially from 21.5 R_o to the Sun centre). It shows that the MESSENGER and STEREO-A first connection to the shock takes place at ~22:55 UT when the shock is at respectively ~3.5 and ~3.2 R_o (Cols. (3) and (5)). Thus, the first connection to the shock occurs at least ~27 minutes before the estimated particle solar release time at ~23:22 UT. If the earlier 3D reconstruction is considered (Sect. 4.5), this first connection time for MES-SENGER and STEREO-A would be at ~22:42 UT, ~40 minutes before the estimated particle solar release time. The MESSENGER cobpoint is located very close to the shock nose and the STEREO-A cobpoint is longitudinally separated by ~23° east from the CME-driven shock nose (Col. (7)).

As STEREO-B is not connected to the shock in the corona, the ENLIL simulated time-dependent CME-driven shock connectivity outside 21.5 R_{\odot} might help to understand the SEP intensity profile at STEREO-B. The location of the simulated shock is identified as speed increases by more than 20 km/s compared to the ambient simulation along magnetic field lines connected to the observer. To derive a particular shock position in the shock files, a background run without that particular CME is subtracted from the run with that particular CME (Bain et al. 2016). For example, to get the position of the shock associated with CME 1, the run of background solar wind is subtracted from the run with background solar wind and CME 1. To get the shock position associated with CME 2, the run of background solar wind and CME 1 is subtracted from the run with background solar wind, CME 1, and CME 2. A general limitation of the model regarding the shock simulation is that the CME input parameters are derived from the fit of the CME LE, but the shock itself may be wider (Bain et al. 2016). Based on the lowresolution run²², STEREO-B becomes magnetically connected to the shock for the first time at a radial distance of 0.63 au, on August 21 at 22:01 UT, which corresponds to the first non-zero numbers in the shock parameters file from the simulation. Figure 13 (b) shows the configuration of heliosphere at this time. Thus, STEREO-B's first connection to the shock occurs within a few hours of the time at which STEREO/LET observed an increase in protons at pitch angle 0° (left panel in Fig. 4), which starts around August 21 at 05:00 UT. Due to the positive magnetic field polarity of this period, the particle increase observed at pitch angle 0° corresponds to particles propagating away from the Sun along the IMF lines.

6. Summary and discussion

On 2013 August 19, a large widespread SEP event was observed by STEREO-A, STEREO-B, the L1 spacecraft, MES-SENGER, and Mars Odyssey, spanning a longitudinal range of 222° in the ecliptic plane, where >2 MeV electron and >50 MeV proton intensity enhancements were observed by the first three spacecraft. The single solar AR (named AR6) associated with the particle enhancements observed in all these spacecraft, from where a CME was ejected, was located near the far-side central meridian from Earth's perspective. The SEP intensity profiles seen by MESSENGER and STEREO-A were very different, although the longitudinal separation between the spacecraft was only 15°, and the footpoint longitudinal separations to AR6

²² https://ccmc.gsfc.nasa.gov/database_SH/Laura_

²³ https://ccmc.gsfc.nasa.gov/database_SH/Laura_

Rodriguez-Garcia_093020_SH_2.php (medium resolution)

were $\Delta\phi=3^{\circ}$ and $\Delta\phi=25^{\circ}$, respectively. MESSENGER observed a prompt steep rise, and STEREO-A observed a delayed and gradual increase. STEREO-B and L1, poorly connected to AR6 ($\Delta\phi=102^{\circ}$ and $\Delta\phi=-138^{\circ}$, respectively) presented time profiles similar to those of STEREO-A, but with more gradual and delayed flux increases. Table 5 summarizes the timeline for the SEP event and the signatures of the associated solar eruption. The wide ($\sim121^{\circ}$) and fast ($\sim1149 \text{ km s}^{-1}$) CME associated with this unusual widespread SEP event was ejected in two stages. The second and main eruption, which occurred at $\sim22:10$ UT, is taken as the time reference for the timeline.

The CME-driven shock was observed for the first time at ~22:35 UT in COR1, with a resulting speed of ~1160 km s⁻¹, based on the spheroid model. S/WAVES observed type II radio bursts starting at ~22:28 UT that persisted for ~8 hours, and whose frequency drift yields a linear fit speed of 1080 km s⁻¹, very similar to the 3D reconstructed CME-driven shock speed. Groups of type III bursts were observed from $\sim 22:28$ to $\sim 00:00$ UT, extending to low frequencies, which suggests that electrons are being lost to the IP medium through open magnetic field lines. This type III emission was relatively intense at low frequencies and was more prominent around ~23:00 UT (second group of emissions), when the type II bursts were more intense too. These type III bursts seem to originate from the coronal-IP type II emission (although there is no higher-frequency information to confirm it) and they might only be associated with the presence of the shock (SA type III). We note that these groups of type III radio emission are similar to what Cane et al. (2002) called type III-1, which could be related with a signature of flarerelated acceleration following the departure of the CME. However, the scenario where the type III bursts are shock-accelerated seems to be reasonable in this unusual SEP event. This is based on the timing of the type III emissions, starting at the same time of the type II emissions and 60 minutes (first group) to 30 minutes (second group) before the post-eruptive arcade started (with no type III emission related with the flare, as detailed below), the long electron rise times observed by SEPT, the hard spectral indices in the electron spectra observed by both STEREO, and the presence of MeV electrons (as shown in Fig. 2). This scenario is corroborated by the statistical study carried by Dresing et al. (2020), where the authors propose that electron events with the above characteristics cannot be explained by a pure flare scenario. These events require another acceleration mechanism that involves a prolonged particle injection. Thus, the CME-driven shock might play an important role in accelerating the electrons observed in this SEP event.

Accompanying the CME eruption, an M-class flare was seen as a post-eruptive arcade starting at ~23:30 UT and with a peak of intensity at ~01:00 UT. The EUVI light curves do not clearly reveal a flare impulsive phase, and there is no type III emission related to the flare, which suggests that there is little or no particle release or acceleration from the flare site. The M-class flare is too late to explain the particle acceleration in this unusual SEP event, as the estimated particle solar release (shifted) time is ~23:22 UT. This result also reflects that of Kahler et al. (2000), who provided evidence against the possibility of solar post-eruptive arcade contributions to gradual SEP events. We note that although there is only 8 minutes difference between the flare onset and solar particle release time, the latter is based on MESSENGER observations, and should be considered as an upper limit due to the anti-sunward direction pointing of the instrument. Thus, the observations support that the likely solar source associated with the widespread event is the shock driven by the CME.

The first spacecraft to observe a particle enhancement was MESSENGER ($\Delta \phi = 3^{\circ}$), already connected to the shock nose area before ~22:42 UT (based on the earlier 3D shock reconstruction). The quasi-relativistic electron onset was observed at \sim 23:27 UT (shifted by light propagation time to 1 au in order to compare with electromagnetic observations). The 59-minute delay with respect to the weak supercritical shock wave formation, as indicated by the type II burst (Claßen & Mann 1998) observed at ~22:28 UT and corresponding to a height of ~2.5 R_{\odot} , could be due to a combination of factors. Firstly, the connection of the spacecraft to the shock is estimated to occur later (~22:42 UT); secondly, the shock might not efficiently accelerate energetic particles until it is strong enough (Kouloumvakos et al. 2019). The last factor is the anti-Sun pointing of MESSENGER instrument. At ~23:22 UT, which corresponds to the estimated release time of particles (shifted to 1 au), the shock heliocentric height is ~5 R_{\odot} (using ENLIL simulated magnetic field lines). This height agrees (at the higher end of the range) with the statistical study on shock properties associated with >50 MeV proton SEP events during solar cycle 24, carried out by Kouloumvakos et al. (2019). They conclude that shock supercriticality is reached between ~1.2 and ~6 R_{\odot} , and the solar particle release times tend to occur from a few minutes up to ~80 minutes after the shock waves become supercritical for well-connected spacecraft ($\Delta \phi < 70^{\circ}$). Thus, in this scenario, where the shock is the main source of the particle acceleration, the time between the first and second group of type III bursts could be related to the time the shock needs to reach supercriticality. We note that at the time of the second group of type III bursts, the emission is enhanced and coincides with more prominent type II bursts starting at ~5 MHz, which corresponds to a heliocentric height of ~4.5 R_☉.

The second spacecraft to observe an increase in particles was STEREO-A ($\Delta \phi = 25^{\circ}$), already connected to an area $\sim 23^{\circ}$ east from the shock nose, before $\sim 22:42$ UT (based on the earlier 3D shock reconstruction). The onset of the relativistic electron (0.7-1.4 MeV) was estimated at ~00:15 UT, which corresponds to a 107-minute delay with respect to the type II-first group of SA type III radio bursts. Richardson et al. (2014) examined the connection-angle dependence of electron and proton delays, derived from observed particle onset times during solar cycle 24 from STEREO and near-Earth spacecraft, relative to the onset of the associated type III radio emissions. If we compare the 107-minute delay with the Richardson et al. (2014) study, we find that STEREO-A relativistic electron onset delay is unusually longer. We note that both MESSENGER and STEREO-A are simultaneously connected to the shock at ~22:42 UT, but at two different shock areas, when the shock is at $\sim 3.0 \text{ R}_{\odot}$ (based on the earlier 3D shock reconstruction and ENLIL model). Thus, assuming that the CME-driven shock is the main source of the particle acceleration, our results indicate that the unusual delay observed in STEREO-A electron onset could be the result of three factors (the first two closely related to each other): firstly, the time the shock needs to become supercritical (as explained above for MESSENGER); secondly, the difference in the cobpoint location between MESSENGER and STEREO-A, whose cobpoint is separated $\sim 23^{\circ}$ east from the nose area; and thirdly, significant particle scattering between the locations of Mercury, at 0.33 au, and STEREO-A, at 0.97 au (in addition to the possible delay due to the larger heliocentric distance of STEREO-A). This is based on the prompt and sharp particle enhancement observed by MESSENGER, the delayed and gradual particle flux increase observed by STEREO-A, the low anisotropies observed by STEREO-A, and the long effective path length derived from

STEREO-A velocity dispersion analysis. If we associated the second group of type III bursts to the SEP event, considered the time the shock would start accelerating energetic particles, the delay would be reduced to 74 minutes. However, this delay would be still considered longer in comparison with the statistical study of Richardson et al. (2014), probably due to the scattering present in the particle transport to STEREO-A location.

The last spacecraft to observe an enhancement in particles are the L1 spacecraft and STEREO-B. SOHO ($\Delta \phi$ =-138°), observing lower intensities for all energies and species than STEREO-B ($\Delta \phi = 102^{\circ}$), seems to present earlier onsets for relativistic electrons than STEREO-B. The gradual increase, which can influence the onset determination (given with 60-minute uncertainty for both locations), and the different instrument designs could be behind the earlier onsets observed at SOHO. Although SOHO/EPHIN observed a higher preceding SEP background at some energies than STEREO-B, SOHO/EPHIN has a much larger geometric factor and a wider field of view. It is also equipped with a lateral anti-coincidence, resulting in much lower instrumental background than STEREO/HET (Fig. 3 in Richardson et al. 2014) and better statistics. STEREO-B relativistic (0.7-1.4 MeV) electron flux started to rise at ~03:00 UT when the spacecraft was not yet connected to the shock along the IP magnetic field lines (based on the 3D shock reconstruction), which corresponds to 272-minute delay from the type II-first group of SA type III radio bursts. This could mean either that the propagating coronal shock kinematics were not well modelled (although there is good agreement between type II radio burst drift and 3D modelled shock height evolution, as presented in Fig. 10), or that the cross-field transport, supported by the low anisotropies and delayed rising fluxes observed by STEREO-B, might be a key factor in the particle propagation, and there is no need to be magnetically connected to the shock in order to observe a SEP event (e.g. Dresing et al. 2012; Laitinen et al. 2013; Dresing et al. 2014; Dröge et al. 2016; Kollhoff et al. 2021). The L1 spacecraft, with a relativistic (0.25-0.70 MeV) electron onset at ~01:00 UT (152-minute delay to the first group of type III bursts), observed an even more gradual increase. Because they were not connected at any time with the shock (based on the 3D shock reconstruction and ENLIL modelling), the low anisotropies and the slow and delayed rising fluxes suggest that the observations are most likely explained in terms of perpendicular diffusion (e.g. Qin & Wang 2015; Strauss et al. 2017).

The acceleration and propagation of particles in this SEP event were also influenced by the interplanetary structures present in the heliosphere. In particular, according to the in situ solar wind data and ENLIL modelling, an ICME was observed at MESSENGER and both STEREO, consistent with the twostage CME erupting from AR6, as no other CMEs within a few hours were ejected from the same AR or surrounding region. We observed an abrupt decrease in intensity following the CMEdriven shock arrival to MESSENGER. STEREO-B observed a late increase in proton anisotropies that occurs 17 hours ahead of the time at which ENLIL modelling predicts the first magnetic connection to the shock on August 21 at ~22:01 UT, when the IP shock was at a heliocentric distance of ~ 0.6 au. For both STEREO spacecraft the acceleration of particles continued for the <10 MeV protons during the evolution of the shock through the heliosphere until its unexpected earlier arrival to STEREO-B (August 22 at 02:09 UT) than to STEREO-A (August 22 at 07:05 UT). STEREO-A observed a particle intensity decrease after the shock arrival, while STEREO-B presented a particle flux increase some hours later, probably related with the spacecraft connecting to a stronger region of the shock closer to the

nose. STEREO-A observed a sudden decrease in particle intensities at the time of the ICME arrival (August 22 at 23:15 UT), and also a transient cavity ~12 hours later, probably related with the closed magnetic field topology of the IP structure. The SEP profile at STEREO-B, that encountered the ICME very close to a flank, seems not to be influenced by the ICME arrival. The detailed analysis of the evolution of the IP shock and ICME, observed at the locations of MESSENGER (0.33 au) and both STEREO spacecraft (1 au), including the evidence of a complex magnetic structure within the ICME, is the focus of a second study (Rodríguez-García et al. 2021).

7. Conclusions

Our main conclusions can be summarized as follows:

- Solar origin: The solar source associated with the widespread SEP event is likely the shock driven by the two-stage CME eruption observed near the far-side central meridian from Earth's perspective, as the M-class flare observed as a post-eruptive arcade is too late to explain the particle acceleration.
- Radio dynamic spectra: Coronal–IP type II radio emission related with the presence of the CME-driven shock is prominent and persists for ~8 hours. Groups of shock-accelerated type III bursts are observed from ~50 minutes before to ~40 minutes after the solar release time of the particles. This type III emission, enhanced around the time of the particle release time, is relatively intense at low (hectometric) frequency, and extends to low frequencies (lower than 100 kHz).
- CME-driven coronal shock: The SEP acceleration might start when the shock heliocentric height is ~5 R_☉. As for other previously observed widespread SEP events, the longitudinal extent of the CME-driven shock does not explain by itself the wide spread of particles in the heliosphere as STEREO-B and L1 are not connected with the coronal shock when the respective particle fluxes start to rise. The shock longitudinal width, figured as to be the equivalent $2xR_{maj}$, is 116° at 21.5 R_☉, while the SEP spread at 1 au is at least 222°.
- Energetic particle propagation: The differences in the SEP time profiles observed by MESSENGER (prompt and sharp) and STEREO-A (delayed and gradual) suggest that particles undergo significant interplanetary scattering between 0.3 and 1 au. The particle enhancement observed at L1 may be attributed to cross-field transport, and this is also the case for STEREO-B, at least until the spacecraft is magnetically connected to the shock when it reaches ~0.6 au, around the time the proton anisotropies presented an increase.

This work illustrates how a two-stage CME drives a single shock in the corona, which accelerates, at a relatively high heliocentric height and with no flare-related contribution, the energetic electrons and protons observed in this unusual SEP event. Regarding the observational differences between the SEP time profiles at MESSENGER (0.33 au) and STEREO-A (1 au), solar missions such as Parker Solar Probe (Fox et al. 2016), launched in August 2018, or Solar Orbiter (Müller et al. 2020; Zouganelis et al. 2020), launched in February 2020, will be crucial for a better understanding of the particle acceleration and transport in the inner heliosphere. The new era of combined in situ and remote-sensing multi-point observations has just started (e.g. Kollhoff et al. 2021), using new data from different instruments like the Energetic Particle Detector (EPD, Rodríguez-Pacheco et al. 2020) on board Solar Orbiter. Acknowledgements. The authors thank an anonymous referee for a detailed review that improved the quality of the manuscript. The UAH team acknowledges the financial support by the Spanish Ministerio de Ciencia, Innovación y Universidades FEDER/MCIU/AEI Projects ESP2017-88436-R and PID2019-104863RB-I00/AEI/10.13039/501100011033. LRG is also supported by the European Space Agency, under the ESA/NPI program, and acknowledges the support from Anik De Groof, Andrew Walsh, Olivier Witasse, Georgina Graham, Ignacio Cernuda Cangas and Francisco Espinosa Lara. FCM acknowledges the financial support by the Spanish MINECO-FPI-2016 predoctoral grant. LB acknowledges the support from the NASA program NNH17ZDA001N-LWS (Awards Nr. 80NSSC19K1261 and 80NSSC19K1235). ND acknowledges financial support by DLR under grant 50OC1702. NVN's work has been supported by NASA contracts 80NSSC18K1126 and 80NSSC20K0287. MD acknowledges support by the Croatian Science Foundation under the project IP-2020-02-9893 (ICOHOSS). LKJ is supported by NASA's Living with a Star and Heliophysics Supporting Research programs. The authors acknowledge the different Wind, ACE, SOHO STEREO, MESSENGER and Mars Odyssey instrument teams, and the STEREO and ACE science centers for providing the data used in this paper. In particular, we would like to thank Jean-Pierre Wuelser, Bill Thompson and Nathan Rich for their help regarding STEREO/SECCHI data. MESSENGER data were downloaded from the Planetary Data System. In addition, this paper uses data from the Heliospheric Shock Database, generated and maintained by the University of Helsinki, and from GONG, managed by the National Solar Observatory. The radio spectrograms used in this study are provided by the Observatoire de Paris-Meudon (http://secchirh.obspm.fr/). ENLIL simulation results have been provided by the CCMC at NASA Goddard Space Flight Center (GSFC) through their public Runs on Request system (http://ccmc.gsfc. nasa.gov; run IDs Laura_Rodriguez-Garcia_093020_SH_1, Laura_Rodriguez-Garcia_093020_SH_2). The WSA model was developed by N. Arge, currently at GSFC, and the ENLIL Model was developed by D. Odstrcil, currently at George Mason University.

ORCID iDs

- Laura Rodríguez-García 💿
- https://orcid.org/0000-0003-2361-5510
- Raúl Gómez-Herrero D https://orcid.org/0000-0002-5705-9236 Ioannis Zouganelis b https://orcid.org/0000-0003-2672-9249 Laura Balmaceda D https://orcid.org/0000-0003-1162-5498 Teresa Nieves-Chinchilla 🗈

https://orcid.org/0000-0003-0565-4890

- Nina Dresing () https://orcid.org/0000-0003-3903-4649
- Nariaki Nitta D https://orcid.org/0000-0001-6119-0221
- Fernando Carcaboso Morales 💿
- https://orcid.org/0000-0003-1758-6194
- Luiz Fernando Guedes dos Santos 🗈
- https://orcid.org/0000-0001-5190-442X
- Lan Jian (https://orcid.org/0000-0002-6849-5527
- Leila Mays (http://orcid.org/0000-0001-9177-8405
- David R Williams David Javier Rodríguez-Pacheco 🗈
- https://orcid.org/0000-0002-4240-1115

References

- Acuña, M. H., Curtis, D., Scheifele, J. L., et al. 2008, Space Sci. Rev., 136, 203 Anderson, B. J., Acuña, M. H., Lohr, D. A., et al. 2007, Space Sci. Rev., 131, 417
- Andrews, G. B., Zurbuchen, T. H., Mauk, B. H., et al. 2007, Space Sci. Rev., 131, 523
- Arge, C. N., Luhmann, J. G., Odstrcil, D., Schrijver, C. J., & Li, Y. 2004, Journal of Atmospheric and Solar-Terrestrial Physics, 66, 1295
- Arge, C. N. & Pizzo, V. J. 2000, J. Geophys. Res., 105, 10465
- Bain, H. M., Mays, M. L., Luhmann, J. G., et al. 2016, The Astrophysical Journal, 825, 1
- Bougeret, J.-L., Goetz, K., Kaiser, M., et al. 2008, Space Science Reviews, 136, 487
- Bougeret, J. L., Kaiser, M. L., Kellogg, P. J., et al. 1995, Space Science Reviews, 71, 231
- Boynton, W. V., Feldman, W. C., Mitrofanov, I. G., et al. 2004, Space Sci. Rev., 110.37
- Brueckner, G. E., Howard, R. A., Koomen, M. J., et al. 1995, Solar Physics, 162, 357
- Article number, page 22 of 23

- Cane, H. V., Erickson, W. C., & Prestage, N. P. 2002, Journal of Geophysical Research (Space Physics), 107, 1315
- Cane, H. V. & Lario, D. 2006, Space Sci. Rev., 123, 45
- Cane, H. V., McGuire, R. E., & von Rosenvinge, T. T. 1986, ApJ, 301, 448
- Cane, H. V., Reames, D. V., & von Rosenvinge, T. T. 1988, J. Geophys. Res., 93, 9555
- Cane, H. V., Stone, R. G., & Fainberg, J. 1981, Geophys. Res., 8, 1285
- Chi, Y., Shen, C., Luo, B., Wang, Y., & Xu, M. 2018, Space Weather, 16, 1960 Chollet, E. E., Mewaldt, R. A., Cummings, A. C., et al. 2010, Journal of Geophysical Research (Space Physics), 115, A12106
- Claßen, H. T. & Mann, G. 1998, Radiophysics and Quantum Electronics, 41, 53
- Cliver, E. W., Thompson, B. J., Lawrence, G. R., et al. 2005, in International Cosmic Ray Conference, Vol. 1, 29th International Cosmic Ray Conference (ICRC29), Volume 1, 121
- Desai, M. & Giacalone, J. 2016, Living Reviews in Solar Physics, 13, 3
- Dodson, H. W. & Hedeman, E. R. 1969, Sol. Phys., 9, 278
- Domingo, V., Fleck, B., & Poland, A. I. 1995, Sol. Phys., 162, 1
- Dresing, N., Effenberger, F., Gómez-Herrero, R., et al. 2020, ApJ, 889, 143
- Dresing, N., Gómez-Herrero, R., Heber, B., et al. 2014, Astrophysical Journal,
- 567, A27 Dresing, N., Gómez-Herrero, R., Klassen, A., et al. 2012, Solar Physics, 281, 281
- Dröge, W., Kartavykh, Y. Y., Dresing, N., & Klassen, A. 2016, ApJ, 826, 134
- Dumbović, M., Guo, J., Temmer, M., et al. 2019, ApJ, 880, 18
- Erickson, W. C. 1997, PASA, 14, 278
- Forbush, S. E. 1946, Physical Review, 70, 771
- Fox, N. J., Velli, M. C., Bale, S. D., et al. 2016, Space Sci. Rev., 204, 7
- Galvin, A. B., Kistler, L. M., Popecki, M. A., et al. 2008, The Plasma and Suprathermal Ion Composition (PLASTIC) Investigation on the STEREO Observatories (New York, NY: Springer New York), 437-486
- Garcia, H. A. 1994, Sol. Phys., 154, 275
- Gloeckler, G., Cain, J., Ipavich, F. M., et al. 1998, Space Sci. Rev., 86, 497
- Gold, R. E., Krimigis, S. M., Hawkins, S. E., I., et al. 1998, Space Sci. Rev., 86,
- 541 Gómez-Herrero, R., Dresing, N., Klassen, A., et al. 2015, Astrophysical Journal, 799.55
- Gómez-Herrero, R., Klasses, A., Heber, B., et al. 2007, in ESA Special Publication, Vol. 641, Second Solar Orbiter Workshop, 63
- Hapgood, M. A. 1992, Planet. Space Sci., 40, 711
- Harvey, J. W., Hill, F., Hubbard, R. P., et al. 1996, Science, 272, 1284
- He, H. Q., Qin, G., & Zhang, M. 2011, ApJ, 734, 74
- Howard, R. A., Moses, J. D., Vourlidas, A., et al. 2008, Space Sci. Rev., 136, 67 Jebaraj, I. C., Magdalenić, J., Podladchikova, T., et al. 2020, A&A, 639, A56
- Jian, L. K., Luhmann, J. G., Russell, C. T., & Galvin, A. B. 2019, Sol. Phys., 294, 31
- Jian, L. K., Russell, C. T., Luhmann, J. G., & Galvin, A. B. 2018, ApJ, 855, 114 Kahler, S. W., McAllister, A. H., & Cane, H. V. 2000, ApJ, 533, 1063
- Kahler, S. W. & Vourlidas, A. 2013, ApJ, 769, 143
- Kaiser, M. L., Kucera, T. A., Davila, J. M., et al. 2008, Space Sci. Rev., 136, 5
- Kallenrode, M. B. 2003, Journal of Physics G Nuclear Physics, 29, 965
- Kallenrode, M.-B. & Wibberenz, G. 1997, J. Geophys. Res., 102, 22311
- Kallenrode, M. B., Wibberenz, G., Kunow, H., et al. 1993, Sol. Phys., 147, 377
- Kay, C., Mays, M. L., & Verbeke, C. 2020, Space Weather, 18, e02382
- Kihara, K., Huang, Y., Nishimura, N., et al. 2020, ApJ, 900, 75 Kocharov, L. & Torsti, J. 2002, Sol. Phys., 207, 149
- Kollhoff et al. 2021, in preparation
- Kouloumvakos, A., Rouillard, A. P., Wu, Y., et al. 2019, ApJ, 876, 80
- Kwon, R.-Y., Zhang, J., & Olmedo, O. 2014, ApJ, 794, 148
- Laitinen, T., Dalla, S., & Marsh, M. S. 2013, ApJ, 773, L29
- Laitinen, T., Huttunen-Heikinmaa, K., Valtonen, E., & Dalla, S. 2015, ApJ, 806, 114
- Lario, D., Aran, A., Gómez-Herrero, R., et al. 2013, Astrophysical Journal, 767, 41
- Lario, D., Kallenrode, M. B., Decker, R. B., et al. 2006, ApJ, 653, 1531
- Lario, D. & Karelitz, A. 2014, Journal of Geophysical Research (Space Physics), 119, 4185
- Lario, D., Kwon, R.-Y., Richardson, I. G., et al. 2017, Astrophysical Journal, 838.51
- Lario, D., Kwon, R. Y., Vourlidas, A., et al. 2016, ApJ, 819, 72
- Lario, D., Raouafi, N. E., Kwon, R. Y., et al. 2014, Astrophysical Journal, 797
- Leblanc, Y., Dulk, G. A., & Bougeret, J.-L. 1998, Sol. Phys., 183, 165
- Lee, C. O., Arge, C. N., Odstrčil, D., et al. 2013, Sol. Phys., 285, 349
- Lemen, J. R., Title, A. M., Akin, D. J., et al. 2012, Sol. Phys., 275, 17
- Luhmann, J. G., Curtis, D. W., Schroeder, P., et al. 2008, Space Sci. Rev., 136,
- 117
- Mäkelä, P., Gopalswamy, N., Akiyama, S., Xie, H., & Yashiro, S. 2015, ApJ, 806, 13
- Mays, M. L., Taktakishvili, A., Pulkkinen, A., et al. 2015, Sol. Phys., 290, 1775
- McComas, D. J., Bame, S. J., Barker, P., et al. 1998, Space Sci. Rev., 86, 563

Mewaldt, R. A., Cohen, C. M. S., Cook, W. R., et al. 2008, Space Sci. Rev., 136, 285

- Müller, D., St. Cyr, O. C., Zouganelis, I., et al. 2020, A&A, 642, A1
- Müller-Mellin, R., Böttcher, S., Falenski, J., et al. 2008, Space Sci. Rev., 136, 363
- Müller-Mellin, R., Kunow, H., Fleißner, V., et al. 1995, Sol. Phys., 162, 483
- Neugebauer, M. 2013, Space Sci. Rev., 176, 125
- Nitta, N. V., Aschwanden, M. J., Boerner, P. F., et al. 2013, Sol. Phys., 288, 241 Odstrcil, D. 2003, Advances in Space Research, 32, 497
- Odstrcil, D., Riley, P., & Zhao, X. P. 2004, Journal of Geophysical Research (Space Physics), 109, A02116
- Odstrčil, D., Smith, Z., & Dryer, M. 1996, Geophys. Res. Lett., 23, 2521 Olmedo, O., Gopalswamy, N., Xie, H., et al. 2013, in AGU Fall Meeting Abstracts, Vol. 2013, SH13A-2033
- Paassilta, M., Papaioannou, A., Dresing, N., et al. 2018, Sol. Phys., 293, 70
- Papaioannou, A., Malandraki, O. E., Dresing, N., et al. 2014, A&A, 569, A96
- Papaioannou, A., Sandberg, I., Anastasiadis, A., et al. 2016, Journal of Space Weather and Space Climate, 6, A42
- Pesnell, W. D., Thompson, B. J., & Chamberlin, P. C. 2012, Sol. Phys., 275, 3 Pohjolainen, S., van Driel-Gesztelyi, L., Culhane, J. L., Manoharan, P. K., & Ponjolainen, S., van Dhel-Geszlelyi, L., Cuinane, J. L., Manonaran, P. K., & Elliott, H. A. 2007, Sol. Phys., 244, 167
 Qin, G. & Wang, Y. 2015, ApJ, 809, 177
 Reames, D. V. 1999, Space Sci. Rev., 90, 413
 Reames, D. V. 2013, Space Sci. Rev., 175, 53
 Reames, D. V., Barbier, L. M., & Ng, C. K. 1996, ApJ, 466, 473
 Richardson, I. G. & Cane, H. V. 1996, J. Geophys. Res., 101, 27521
 Richardson, I. G., Cane, H. V., 2010, Sol. Phys., 264, 189
 Richardson, I. G., Cane, H. V., & von Rosenvinge, T. T. 1991, J. Geophys. Res., 06, 7753

- 96.7853 Richardson, I. G., von Rosenvinge, T. T., Cane, H. V., et al. 2014, Solar Physics,
- 289.3059
- Rodríguez-Pacheco, J., Wimmer-Schweingruber, R. F., Mason, G. M., et al. 2020, A&A, 642, A7
- Rodríguez-García et al. 2021, in preparation
- Saito, K., Makita, M., Nishi, K., & Hata, S. 1970, Annals of the Tokyo Astronomical Observatory, 12, 51
- Sanahuja, B., Lario, D., & Heras, A. M. 1995, in International Cosmic Ray Conference, Vol. 4, International Cosmic Ray Conference, 361
- Saunders, R. S., Arvidson, R. E., Badhwar, G. D., et al. 2004, Space Sci. Rev., 110, 1
- Smith, C. W., L'Heureux, J., Ness, N. F., et al. 1998, Space Sci. Rev., 86, 613 Solomon, S. C., McNutt, R. L., Gold, R. E., & Domingue, D. L. 2007, Space Sci. Rev., 131, 3
- Stone, E. C., Frandsen, A. M., Mewaldt, R. A., et al. 1998, Space Sci. Rev., 86,
- Strauss, R. D., Dresing, N., Kollhoff, A., & Brüdern, M. 2020, ApJ, 897, 24
- Strauss, R. D. T., Dresing, N., & Engelbrecht, N. E. 2017, ApJ, 837, 43 Susino, R., Bemporad, A., & Mancuso, S. 2015, ApJ, 812, 119
- Szabo, A. 2015, Handbook of Cosmic Hazards and Planetary Defense, 141 Thernisien, A. 2011, ApJS, 194, 33
- Thernisien, A., Vourlidas, A., & Howard, R. A. 2009, Solar Physics, 256, 111 Thernisien, A. F. R., Howard, R. A., & Vourlidas, A. 2006, ApJ, 652, 763
- Thompson, B. J., Plunkett, S. P., Gurman, J. B., et al. 1998, Geophys. Res. Lett., 25, 2465
- Thompson, W. T. 2006, A&A, 449, 791
- Torsti, J., Kocharov, L., Teittinen, M., et al. 1999, J. Geophys. Res., 104, 9903 Torsti, J., Valtonen, E., Lumme, M., et al. 1995, Sol. Phys., 162, 505
- Vainio, R., Valtonen, E., Heber, B., et al. 2013, Journal of Space Weather and Space Climate, 3, A12 von Rosenvinge, T. T., Reames, D. V., Baker, R., et al. 2008, Space Sci. Rev.,
- 136, 391
- Vourlidas, A., Howard, R. A., Esfandiari, E., et al. 2011, ApJ, 730, 59 Vourlidas, A., Subramanian, P., Dere, K. P., & Howard, R. A. 2000, ApJ, 534, 456
- Vršnak, B., Magdalenić, J., & Zlobec, P. 2004, A&A, 413, 753 Wibberenz, G. & Cane, H. V. 2006, Astrophysical Journal, 650, 1199, provided by the SAO/NASA Astrophysics Data System
- Wold, A. M., Mays, M. L., Taktakishvili, A. r., et al. 2018, Journal of Space Weather and Space Climate, 8, A17
- Wuelser, J.-P., Lemen, J. R., Tarbell, T. D., et al. 2004, in Society of Photo-Optical Instrumentation Engineers (SPIE) Conference Series, Vol. 5171, Telescopes and Instrumentation for Solar Astrophysics, ed. S. Fineschi & M. A. Gummin. 111–122
- Xie, H., Mäkelä, P., St. Cyr, O. C., & Gopalswamy, N. 2017, Journal of Geophysical Research (Space Physics), 122, 7021 Xie, H., St. Cyr, O. C., Mäkelä, P., & Gopalswamy, N. 2019, Journal of Geo-
- physical Research (Space Physics), 124, 6384
- Yashiro, S., Gopalswamy, N., Michalek, G., et al. 2004, Journal of Geophysical Research (Space Physics), 109, A07105
- Zeitlin, C., Boynton, W., Mitrofanov, I., et al. 2010, Space Weather, 8, S00E06 Zhao, L., Li, G., Zhang, M., et al. 2019, ApJ, 878, 107
- Zouganelis, I., De Groof, A., Walsh, A. P., et al. 2020, A&A, 642, A3